# PART 2 — FINDINGS AND RECOMMENDATIONS

The Review's Terms of Reference require an assessment of Australia's competition policy, laws and institutions to determine whether they remain fit for purpose, especially in light of the opportunities and challenges facing Australia into the foreseeable future.

In this Part, we summarise the findings of this assessment and set out recommendations to address deficiencies the Panel has identified.

Chapter 2 presents the Panel's recommendations for priority areas of reform in competition policy.

These are informed by a set of competition principles attuned to the challenges and opportunities likely to face the Australian economy in coming decades. A key lesson from the National Competition Policy (NCP) experience is the importance of an agreed framework, which can then be applied by governments in their own jurisdictions and adapted to local conditions as necessary.

A further lesson from NCP is that all reform initiatives cannot be progressed simultaneously. The Panel recognises the importance of assigning priorities to reform initiatives so that those with the greatest potential benefit to Australians are progressed first. Moreover, priorities will change as technology changes — for instance, the development of the National Broadband Network (NBN) and mobile telephony infrastructure have meant that access to the 'unbundled local loop' (i.e., the copper network) is a less significant issue than it was in 1995.

Competition policy reforms most likely to generate large net benefits are those that: benefit a sizeable part of the economy or have deep links to other sectors; remove a significant barrier to competition; or subject activities with significant government involvement to greater contestability and consumer choice.

Chapter 3 outlines the Panel's recommendations for changes to the *Competition and Consumer Act 2010* (CCA).

The Panel has viewed reform of the CCA through the lens of fitness for purpose. In some areas, we recommend substantive changes to the way the law is drafted. In other areas, our recommended changes go to clarifying and simplifying the law.

On some issues, the Panel finds the law itself fit for purpose but shares concerns expressed by stakeholders, especially small business, about access to remedies under the law.

Chapter 4 outlines the Panel's recommendations on the **institutional structures** most likely to sustain enduring reform.

Like the Hilmer Review, we recognise that policy reform will only gain and sustain momentum if it is supported by all jurisdictions.

Australia has been well served by its competition policy institutions, yet this is not sufficient reason to retain the framework in its current form. The flagging momentum of competition reform points to the need for reinvigoration through strong institutional frameworks.

The Panel has identified a clear gap in the competition framework: an institution is needed to advocate for competition reform and to oversee the implementation of reforms instituted by governments in the wake of this Review.

Chapter 5 outlines the Panel's recommendations relating to concerns that **small business** has raised with us.

Access to remedies has been a roadblock for many small businesses, and the Panel finds that access should be improved. We recommend that the collective bargaining framework should be enhanced and made more flexible. We also make recommendations on competitive neutrality and regulations that can restrict the way small businesses operate.

Chapter 6 highlights recommendations made in other parts of this Report addressing issues raised with the Panel that relate to **retail markets**, particularly supermarkets.

Chapter 7 presents the Panel's views on the best method to **implement** a national competition reform agenda. We also recommend economic modelling of the package of recommendations in this Review, which will inform governments' discussions of policy proposals they will pursue.

## 2 COMPETITION POLICY

## 2.1 A SET OF COMPETITION PRINCIPLES

As originally crafted, the National Competition Policy (NCP) reflected the challenges Australia faced more than 20 years ago. The focus of the NCP reforms was on exposing some previously sheltered activities to competition and applying a more national approach to competition issues.

The six elements of competition policy identified in the Hilmer Review<sup>18</sup> were:

- limiting anti-competitive conduct of firms;
- reforming regulation which unjustifiably restricts competition;
- reforming the structure of public monopolies to facilitate competition;
- providing third-party access to certain facilities that are essential for competition;
- · restraining monopoly pricing behaviour; and
- fostering 'competitive neutrality' between government and private businesses when they compete.

The Panel endorses competition policy that focuses on making markets work in the long-term interests of consumers. Legislative frameworks should continue to limit anti-competitive conduct of firms. However, through its commercial arrangements entered into with market participants, the Crown (whether in right of the Commonwealth, state, territory or local governments) also has the potential to harm competition.

The Panel therefore concludes that the anti-competitive conduct provisions of the *Competition and Consumer Act 2010* (CCA) should cover government activities that have a trading or commercial character.

Moreover, the Crown's capacity to enhance or harm competition reaches beyond the scope of the CCA and includes a range of policies and regulations. In particular, procurement, which ranges from buying goods and services through to public-private partnerships (PPPs) and privatisations, should be designed with competition policy in mind.

The Panel believes that the focus of competition policy should be widened beyond infrastructure public monopolies and government businesses, to encompass the provision of government services more generally.

By promoting user choice and encouraging a diversity of providers, competition policy plays an important role in improving performance in sectors such as human services. Choice and diversity have the potential to improve outcomes for users, especially but not only by stimulating innovation.

Independent regulation can encourage market entry since it provides a level of certainty about the regulatory environment. Similarly, separating the interests of providers from those of funders and

<sup>18</sup> Commonwealth of Australia 1993, *National Competition Policy* (the Hilmer Review), Australian Government Publishing Service, Canberra, page xvii.

regulators encourages accountability, innovation and a level playing field between public and other providers.

The Panel believes that declaration and third-party access to infrastructure should only be mandated when it is in the public interest. The onus of proof should lie with those seeking access to demonstrate that it would promote the public interest rather than on infrastructure owners to demonstrate that access would be contrary to the public interest.

Acknowledging the diverse circumstances of each jurisdiction, the Panel supports the flexibility built into the NCP for the Australian Government and state and territory governments to decide how best to implement competition principles in their jurisdictions. Competition policy should continue to apply explicitly to local government.

Agreeing a set of principles would guide the Australian Government, state, territory and local governments in implementing those aspects of competition policy for which they are responsible. The principles in Recommendation 1 broaden the NCP agenda to include all government services in trade or commerce and promote the role of choice.

In applying these principles the Panel endorses a 'public interest' test as a central tenet of competition policy. The Panel recommends continuing with the NCP public interest test, namely that legislation or government policy should not restrict competition unless:

- the benefits of the restriction to the community as a whole outweigh the costs; and
- the objectives of the legislation or government policy can only be achieved by restricting competition.

#### Recommendation 1 — Competition principles

The Australian Government, state and territory and local governments should commit to the following principles:

- Competition policies, laws and institutions should promote the long-term interests of consumers.
- Legislative frameworks and government policies and regulations binding the public or private sectors should not restrict competition.
- Governments should promote consumer choice when funding, procuring or providing goods and services and enable informed choices by consumers.
- The model for government provision or procurement of goods and services should separate the interests of policy (including funding), regulation and service provision, and should encourage a diversity of providers.
- Governments should separate remaining public monopolies from competitive service elements, and also separate contestable elements into smaller independent business activities.
- Government business activities that compete with private provision, whether for-profit or not-for-profit, should comply with competitive neutrality principles to ensure they do not enjoy a net competitive advantage simply as a result of government ownership.
- A right to third-party access to significant bottleneck infrastructure should be granted where it would promote a material increase in competition in dependent markets and would promote the public interest.
- Independent authorities should set, administer or oversee prices for natural monopoly infrastructure providers.

Applying these principles should be subject to a public interest test, such that legislation or government policy should not restrict competition unless:

- the benefits of the restriction to the community as a whole outweigh the costs; and
- the objectives of the legislation or government policy can only be achieved by restricting competition.

For further detail on competition principles, see Chapter 8.

## 2.2 DETERMINING PRIORITY AREAS FOR REFORM

The Panel recognises the importance of assigning priorities to reform initiatives so that those with the greatest potential benefit to Australians are progressed first.

In determining priority areas for competition policy reform, the Panel has asked five questions:

- Will this reform help the Australian economy adjust to the forces for change identified in Part 1 of this Report?
- Will this reform promote choice, diversity and innovation in markets for private and/or government goods and services?
- Will this reform help to raise productivity growth and hence Australian living standards over time?
- Will this reform stimulate competition or contestability in markets by lowering barriers to entry or exit?
- Will this reform help to complete unfinished business from the original NCP agenda or address specific issues raised in the Review's Terms of Reference?

If the answer to one or more of these questions is 'yes', then the reform is placed on the Panel's priority list. The remaining sections of this chapter present the Panel's recommendations in respect of each of its priority areas for reform.

## 2.3 HUMAN SERVICES

Access to high-quality human services — including health, education and community services — is vital to the lives of all Australians. Good health makes it easier for people to participate in society; education can help put people on a better life pathway; and quality community services, including aged care and disability care and support, can provide comfort, dignity and increased opportunities to vulnerable Australians.

Given the size of the human services sector (which is set to increase further as Australia's population ages), <sup>19</sup> even small improvements will have profound impacts on people's standard of living and quality of life.

The Panel notes that governments are making significant changes across human services sectors, with policies reflecting the unique characteristics of each jurisdiction and the service in question. These changes include a clearer focus on user choice and innovation in service delivery.

As a first step, where governments are involved in human services sectors as a provider, separating the interests of both the regulator and policy-maker (including funding) from the interests of the provider can help to ensure that decisions are made in the best interests of users. Regulation and policy decisions that are independent of government provision can encourage a more certain and stable environment, which can in turn encourage a diversity of new providers.

-

<sup>19</sup> Australian Government 2015, 2015 Intergenerational Report, Australia in 2055, Canberra.

But governments cannot distance themselves from the quality of services delivered to Australians. An ongoing market stewardship function means that governments will retain responsibility for overseeing the impact of policies on users.

The Panel considers that a 'presumption of choice' could have significant benefits in many human services sectors. Putting users in control of the human services they access — either through direct payments, personal budgets, entitlements or choice — drives service providers to become more responsive to individual requirements.

However, the Panel acknowledges that choice is not the only important objective in the area of human services. Equity of access, universal service provision and minimum quality are also important to all Australians.

In considering whether it should recommend change in this area, the Panel does not wish to discourage or crowd out the important contribution that not-for-profit providers and volunteers currently make to the wellbeing of Australians.

Where governments retain some control over the delivery of human services, a diversity of service providers and high-quality outcomes for users can be encouraged through careful commissioning. Governments need to allow room for providers to innovate in response to changing user demands, and to benchmark the performance of providers, credibly threatening to replace those that do not meet the needs of users.

The Panel recognises that some markets will not have sufficient depth to support a number of providers — including, for example, certain services in remote and regional areas. Ensuring access to services and maintaining and improving service quality in these cases increases the emphasis on well-designed benchmarking of services.

The Panel is satisfied that deepening and extending competition policy in human services is a priority reform. Lowering barriers to entry can stimulate a diversity of providers, which expands user choice. Small gains in productivity (driven by competition) in these large and growing sectors of the Australian economy have the potential to deliver large gains across the community.

Reforms in this area can also exert a powerful demonstration effect. If competition produces conspicuous improvements in users' access to and experience of a particular human service, this will strengthen the case for reform across a wider range of government services.

#### Recommendation 2 — Human services

Each Australian government should adopt choice and competition principles in the domain of human services.

Guiding principles should include:

- User choice should be placed at the heart of service delivery.
- Governments should retain a stewardship function, separating the interests of policy (including funding), regulation and service delivery.
- Governments commissioning human services should do so carefully, with a clear focus on outcomes.
- A diversity of providers should be encouraged, while taking care not to crowd out community and volunteer services.
- Innovation in service provision should be stimulated, while ensuring minimum standards of quality and access in human services.

When developing implementation plans, governments can expand on these principles to achieve their goals.

For example, in putting **user choice** at the heart of service delivery, governments should:

- recognise that users are best placed to make choices about the human services they need and design service delivery, wherever possible, to be responsive to those choices;
- recognise that access to quality services will be a prerequisite for effective choice and that accessibility will be particularly important in remote and regional areas;
- ensure that users have access to relevant information to help them exercise their choices, including, where appropriate, feedback from previous users of services;
- in sectors where choice may be difficult, make intermediaries or purchase advisors available to help users make decisions, with policies designed to align the incentives of purchase advisors with the best interests of users;
- ensure that a default option is available for users unable or unwilling to exercise choice;
- lower financial and non-financial switching costs to enable switching wherever possible for
  example, users should not 'lose their place in the queue' if they switch providers, or need to
  undergo further eligibility assessment; and
- offer disadvantaged groups greater assistance in navigating the choices they face through, for example, accessible communications channels that suit their needs.

In undertaking their **stewardship** role, governments should:

- foster a diverse range of service models that best meet the needs of individuals and the broader community;
- co-design markets with human services providers to build on the trust and relationships that already exist between service providers and users;
- separate their interest in policy (including funding) and regulation from provision;
- vest rule-making and regulation with a body independent of government's policy (including funding) role;

- allow funding to follow people's choices; and
- fund community service obligations in a transparent and contestable manner.

#### In **commissioning** human services, governments should:

- encourage careful commissioning decisions that are sensitive and responsive to individual and community needs, and recognise the contribution of community organisations and volunteers;
- ensure that commissioned services are contestable and service providers face credible threats of replacement for poor performance;
- establish targets and benchmarks for service providers based on outcomes, not processes or inputs; and
- offer financial rewards for performance above specified targets.

#### In encouraging a **diversity** of service providers, governments should:

- allow independent regulators to license any provider that meets and maintains prescribed standards, where minimum standards address quality requirements without raising artificial barriers to entry; and/or
- directly commission services with co-ordination and processes that:
  - avoid monopoly providers developing over time; and
  - specify contracts with duration periods that balance the need to afford providers some level of certainty without excluding potential competitors for extended periods of time; and
- in support of their role as market stewards, undertake commissioning that:
  - provides for sufficient information and feedback loops to improve the design and targeting of contracts over time, including by identifying the relative strengths of different types of service provider;
  - recognises the integrated nature of many human services and their joint role in contributing to end-user outcomes, and the relative strengths of different providers in different parts of a co-ordinated service supply chain; and
  - is co-ordinated over time, where possible, maximising opportunities for contracts with overlapping timeframes and supporting a diversity of providers in the market at any point in time.

In encouraging innovation in service delivery, governments should:

- encourage experimental service delivery trials whose results are disseminated via an intergovernmental process; and
- encourage jurisdictions to share knowledge and experience in the interest of continuous improvement.

For further detail on human services, see Chapter 12.

## 2.4 TRANSPORT

## Road transport

Road transport is a major input for business-to-business transactions and, with the rapid growth of online purchases, an increasingly important component of end-point sales to consumers. An efficient road system is also essential for urban and regional access and amenity.

Even small changes in productivity in this sector can cascade through the economy, boosting productivity and output in other sectors. Also, given the size of the road transport sector, enhanced productivity in road transport can deliver large gains to the economy.

However, roads are the least reformed of all infrastructure sectors, with institutional arrangements around funding and provision remaining much the same as they were 20 years ago.

More effective institutional arrangements are needed to promote efficient investment in and usage of roads, and to put road transport on a similar footing with other infrastructure sectors. Lack of proper road pricing leads to inefficient road investment and distorts choices between transport modes, particularly between road and rail freight.

The advent of new technology presents opportunities to improve the efficiency of road transport in ways that were unattainable two decades ago. Road user charges linked to road construction, maintenance and safety should make road investment decisions more responsive to the needs and preferences of road users. As in other network sectors, where pricing is introduced, it should be overseen by an independent regulator.

A critical concern of stakeholders, shared by the Panel, is that road pricing should not be an additional impost on road users. To ensure any reform is revenue-neutral, indirect taxes and charges, such as fuel excise and registration fees, should be reduced as road pricing is introduced. This would make the road sector more like other infrastructure sectors, since road authorities would charge directly and transparently for road use, and allocate the revenue raised to the network's construction and operating costs. Cost-reflective pricing should lead to better road investment decisions, which will make the community and road users better off.

#### Recommendation 3 — Road transport

Governments should introduce cost-reflective road pricing with the aid of new technologies, with pricing subject to independent oversight and revenues used for road construction, maintenance and safety.

To avoid imposing higher overall charges on road users, governments should take a cross-jurisdictional approach to road pricing. Indirect charges and taxes on road users should be reduced as direct pricing is introduced. Revenue implications for different levels of government should be managed by adjusting Australian Government grants to the States and Territories.

## Liner shipping (Part X) and cabotage (coastal shipping and aviation)

The Review's Terms of Reference (3.3.5) require it to consider whether existing exemptions from the competition law and/or historic sector-specific arrangements are still warranted. This includes Part X of the CCA, which exempts international liner cargo shipping from certain competition provisions, including cartel conduct, contracts, arrangements or understandings that affect competition, and exclusive dealing.

Liner shipping is a vital mechanism through which goods cross Australia's borders, both for export and import. Many items moved by sea cannot be transported by air because of their weight or volume. These include not only finished goods but also intermediate inputs for Australian businesses.

The importance of international trade to Australia's economy, and the prospects for stronger growth in trade as Asia develops, focus attention on the need for efficient and competitive marine transportation.

Part X of the CCA allows liner shipping operators to enter into agreements among themselves in relation to the freight rates to be charged, and the quantity and kinds of cargo to be carried, on particular trade routes, and to register those agreements with the Registrar of Liner Shipping (an office created under Part X). Registration confers an exemption from the cartel conduct prohibitions and sections 45 and 47 of the CCA. Although the test for registering a conference agreement under Part X involves assessing the agreement's 'overall benefit' to Australia, it does not expressly require assessing its competitive effects. Also, the test is not assessed by the primary competition regulator, the Australian Competition and Consumer Commission (ACCC), but by the Registrar of Liner Shipping.

No other industry enjoys legislative exemption from Australia's competition laws. This is despite the fact that other industries have similar economic characteristics to the liner shipping industry, particularly the international airline industry. If participants in other industries wish to make agreements that would otherwise contravene the competition law, they are required to seek authorisation from the ACCC.

The Panel therefore considers that Part X should be repealed and the liner shipping industry should be subject to the normal operation of the CCA.

The ACCC should be given power to grant block exemptions (see Recommendation 39 and Section 22.3) for conference agreements that meet a minimum standard of pro-competitive features. For example, conference agreements that co-ordinate scheduling and the exchange of capacity, while allowing confidential individual service contracts and not involving a common conference tariff and pooling of revenues and losses, should be eligible for a block exemption. Other forms of agreement that do not qualify for the block exemption, and thereby risk contravening Part IV provisions, should be subject to individual authorisation.

Repeal of Part X will mean that existing liner shipping agreements will no longer be exempt from the competition law and some may contravene it. A transition period will therefore be needed to establish which agreements qualify for the block exemption and for other agreements to either seek authorisation or be modified if needed to comply with the CCA. The Panel considers a transition period of two years should be sufficient.

The Panel is aware that the Australian Government is undertaking a separate review of coastal shipping regulations but observes that cabotage restrictions raise the cost and administrative complexity of coastal shipping services. The Panel notes that restrictions on air cabotage are stricter than shipping cabotage and that the current blanket restrictions are likely to be inefficient. Cabotage restrictions that are not in the public interest should be removed.

#### Recommendation 4 — Liner shipping

Part X of the CCA should be repealed.

A block exemption granted by the ACCC should be available for liner shipping agreements that meet a minimum standard of pro-competitive features (see Recommendation 39). The minimum standard of pro-competitive features to qualify for the block exemption should be determined by the ACCC in consultation with shippers, their representative bodies and the liner shipping industry.

Other agreements that risk contravening the competition provisions of the CCA should be subject to individual authorisation, as needed, by the ACCC.

Repeal of Part X will mean that existing agreements are no longer exempt from the competition provisions of the CCA. Transitional arrangements are therefore warranted.

A transitional period of two years should allow for the necessary authorisations to be sought and to identify agreements that qualify for the proposed block exemption.

## Recommendation 5 — Cabotage — coastal shipping and aviation

Noting the current Australian Government Review of Coastal Trading, cabotage restrictions on coastal shipping should be removed, unless it can be demonstrated that the benefits of the restrictions to the community as a whole outweigh the costs, and the objectives of the government policy can only be achieved by restricting competition.

The current air cabotage restrictions should be removed for all air cargo as well as passenger services to specific geographic areas, such as island territories and on poorly served routes, unless it can be demonstrated that the benefits of the restrictions to the community as a whole outweigh the costs, and the objectives of the restrictions can only be achieved by restricting competition.

Introducing an air cabotage permit system would be one way of regulating air cabotage services more effectively where necessary.

For further detail on transport, see Section 11.3.

## 2.5 INTELLECTUAL PROPERTY

Disruptive technologies, especially digital technologies, are a pervasive force for change in the Australian economy. New technologies foster innovation, which in turn drives growth in living standards. Access to and creation of intellectual property (IP) will become increasingly important as Australia moves further into the digital age.

Australians are enthusiastic adopters and adapters of new technology. We stand to benefit greatly by exploiting technology to its full extent in our business production processes and as end-consumers. Our IP policy settings should encourage this.

Nevertheless, an appropriate balance must be struck between encouraging widespread adoption of new productivity-enhancing techniques, processes and systems on the one hand, and fostering ideas and innovation on the other. Excessive IP protection can not only discourage adoption of new technologies but also stifle innovation.

Given the influence of Australia's IP rights on facilitating (or inhibiting) innovation, competition and trade, the Panel believes the IP system should be designed to operate in the best interests of Australians.

The Panel therefore considers that Australia's IP rights regime is a priority area for review.

Determining the appropriate extent of IP protection is complex. IP rights can help to break down barriers to entry but, when applied inappropriately, can also reduce exposure to competition and erect long-lasting barriers to entry that fail to serve Australia's interests over the longer term. This risk is especially prevalent in commitments entered into as part of international trade agreements.

The Panel is concerned that Australia has no overarching IP policy framework or objectives guiding changes to IP protection or approaches to IP rights in the context of negotiations for international trade agreements.

## Recommendation 6 — Intellectual property review

The Australian Government should task the Productivity Commission to undertake an overarching review of intellectual property. The Review should be a 12-month inquiry.

The review should focus on: competition policy issues in intellectual property arising from new developments in technology and markets; and the principles underpinning the inclusion of intellectual property provisions in international trade agreements.

A separate independent review should assess the Australian Government processes for establishing negotiating mandates to incorporate intellectual property provisions in international trade agreements.

Trade negotiations should be informed by an independent and transparent analysis of the costs and benefits to Australia of any proposed intellectual property provisions. Such an analysis should be undertaken and published before negotiations are concluded.

The Panel considers it appropriate that commercial transactions involving IP rights, including the assignment and licensing of such rights, be subject to the CCA, in the same manner as transactions involving other property and assets.

Subsection 51(3) of the CCA provides a limited exception from most of the competition law prohibitions for certain types of transactions involving IP. The exception covers conditions in licences or assignments of IP rights in patents, registered designs, copyright, trademarks and circuit layouts where, broadly, the condition relates to products that are the subject of the application of the IP right. The exception does not extend to the prohibitions relating to misuse of market power and resale price maintenance.

It is important to note that subsection 51(3) does not exempt all transactions involving IP rights from competition law; it only exempts certain conditions in a licence or assignment. For example, the transfer of IP rights, whether by licence or assignment, which results in an increase in market power and a consequential substantial lessening of competition is subject to sections 45 and 50; the decision by an IP owner to refuse to license IP rights to another person is subject to the potential application of section 46.

The rationale for excepting conditions in licences or assignments of IP rights is flawed. The rationale assumes that the imposition of conditions in licences and assignments cannot extend the scope of the exclusive rights granted to the IP owner and therefore cannot harm competition (beyond the effect of the original grant of the IP right). In many instances, that will be the case; but in those instances the licence or assignment would not contravene the competition law in any event, making the exception unnecessary. However, in other instances, the assumption will not apply. In fields with multiple and competing IP rights, such as the pharmaceutical or communications industries, cross-licensing arrangements can be entered into to resolve disputes that impose anti-competitive

restrictions on each licensee. The Panel considers that arrangements of this type should be examinable under the competition law.

In most comparable countries, no equivalent to subsection 51(3) exists. None of the US, Canada or Europe provides an exemption from its competition laws for conditions of IP transactions.

The Panel considers that the IP licensing exception in subsection 51(3) of the CCA should be repealed.

However, as is the case with other vertical supply arrangements, IP licences should remain exempt from the per se cartel provisions of the CCA insofar as they impose restrictions on goods or services produced through application of the licensed IP.

IP licensing or assignment arrangements that are at risk of breaching Part IV of the CCA (which covers anti-competitive practices), but which are likely to produce offsetting public benefits, can be granted an exemption from the CCA through the notification or authorisation processes.

In addition, the block exemption power recommended by the Panel (see Recommendation 39) could be used to specify 'safe harbour' licensing restrictions for IP owners.

### Recommendation 7 — Intellectual property exception

Subsection 51(3) of the CCA should be repealed.

For further detail on intellectual property, see Chapter 9.

## 2.6 REGULATORY RESTRICTIONS

The NCP reforms substantially reduced the amount of anti-competitive regulation. Governments made a concerted effort to examine and reform regulation that restricted competition where those restrictions were not in the public interest.

However, the regulation review process, begun under the NCP, has flagged and reinvigoration is now needed. Three areas require governments' attention:

- initiating a new round of regulatory reviews;
- **priority areas for review** (planning and zoning, taxis and ride-sharing, and mandatory product standards); and
- areas for immediate reform action (trading hours, parallel imports and pharmacy).

## A new round of regulatory reviews

Submissions raise many examples of regulatory restrictions on competition, including product standards, taxi licensing, professional and occupational licensing, broadcast media rules, liquor and gambling regulation, private health insurance regulation, agricultural marketing rules and air services restrictions.

Cumulatively, such restrictions can have a significant impact on the economy. Many sectors facing regulatory restrictions supply critical inputs to other business activities. Accordingly, a new round of national regulatory reviews is required. A national approach will provide momentum, impose discipline on all jurisdictions, and foster the emergence of a nationally consistent business regulatory environment.

Rigorous, transparent and independent assessment of whether regulations are in the public interest, with the onus of proof on the party wishing to retain anti-competitive regulation, is important to ensure that regulations serve the long-term interests of consumers. In that vein, the Panel acknowledges submissions expressing concern about excessive deregulation, and accepts that many regulations are essential for other policy reasons. We need *better* regulation rather than no regulation at all.

Opportunities will also arise to examine regulations when reviews are undertaken for other purposes. For example, Australian Government reviews in the communications portfolio<sup>20</sup> should consider the impact of current restrictions on competition in that sector.

Certain activities can be exempted from the operation of the competition law under Part IV of the CCA (apart from the merger laws) through authorisation in Commonwealth, state or territory legislation (subsection 51(1) of the CCA). The Panel believes that such jurisdictional exemptions for conduct that would normally contravene the competition law should be examined to ensure they remain necessary and appropriate in their scope. Any further exemptions should be drafted as narrowly as possible to give effect to their policy intent.

## Recommendation 8 — Regulation review

All Australian governments should review regulations, including local government regulations, in their jurisdictions to ensure that unnecessary restrictions on competition are removed.

Legislation (including Acts, ordinances and regulations) should be subject to a public interest test and should not restrict competition unless it can be demonstrated that:

- the benefits of the restriction to the community as a whole outweigh the costs; and
- the objectives of the legislation can only be achieved by restricting competition.

Factors to consider in assessing the public interest should be determined on a case-by-case basis and not narrowed to a specific set of indicators.

Jurisdictional exemptions for conduct that would normally contravene the competition law (by virtue of subsection 51(1) of the CCA) should also be examined as part of this review, to ensure they remain necessary and appropriate in their scope. Any further exemptions should be drafted as narrowly as possible to give effect to their policy intent.

The review process should be transparent, with highest priority areas for review identified in each jurisdiction, and results published along with timetables for reform.

The review process should be overseen by the proposed Australian Council for Competition Policy (see Recommendation 43) with a focus on the outcomes achieved rather than processes undertaken. The Australian Council for Competition Policy should publish an annual report for public scrutiny on the progress of reviews of regulatory restrictions.

For further detail on regulatory restrictions, see Chapter 10.

See, for example: Australian Government 2014, Spectrum Review, Australian Government Department of Communications, viewed 9 February 2015,

<sup>&</sup>lt;a href="http://www.communications.gov.au/consultation\_and\_submissions/spectrum\_review">http://www.communications.gov.au/consultation\_and\_submissions/spectrum\_review>.</a>

## Priority areas for review

The Panel has identified three priority areas that should be reviewed immediately — planning and zoning rules, taxi regulation and product standards. Across jurisdictions these will vary in their complexity and in their potential benefits, reflecting both the form of the restrictions and the extent of reform previously undertaken.

## Planning and zoning

Land is an important input to the production of goods and services and a source of amenity for consumers. Even small policy improvements in this area could yield large benefits to the economy.

Planning systems by their nature create barriers to entry, diversification or expansion, including through limiting the number, size, operating model and mix of businesses. This can reduce the responsiveness of suppliers to the needs of consumers.

Planning regulations should work in the long-term interests of consumers. They should not restrict competition unless the benefits of the restriction to the community as a whole outweigh the costs, and the objectives of the regulations can only be achieved by restricting competition. Subjecting planning regulations to the public interest test will ensure they do not inappropriately limit entry to markets.

Governments around the country recognise the concerns raised by poorly designed planning and zoning systems, and reviews are either underway or have recently been completed in a number of jurisdictions.

An opportunity exists to ensure that, when undertaking these reviews or implementing their findings, enhanced competition is a central objective. That a number of reviews are already underway also provides the opportunity to compare across jurisdictions to determine best practice as a basis for updating and improving current requirements.

An independent body, such as the proposed Australian Council for Competition Policy (ACCP) (see Recommendation 43), should work with States and Territories to oversee incorporation of competition policy principles in planning and zoning rules. The ACCP should also report to jurisdictions on progress in implementing these principles.

#### Recommendation 9 — Planning and zoning

Further to Recommendation 8, state and territory governments should subject restrictions on competition in planning and zoning rules to the public interest test, such that the rules should not restrict competition unless it can be demonstrated that the benefits of the restriction to the community as a whole outweigh the costs, and the objectives of the rules can only be achieved by restricting competition.

The following competition policy considerations should be taken into account:

- Arrangements that explicitly or implicitly favour particular operators are anti-competitive.
- Competition between individual businesses is not in itself a relevant planning consideration.
- Restrictions on the number of a particular type of retail store contained in any local area is not a relevant planning consideration.
- The impact on the viability of existing businesses is not a relevant planning consideration.
- Proximity restrictions on particular types of retail stores are not a relevant planning consideration.
- Business zones should be as broad as possible.
- Development permit processes should be simplified.
- Planning systems should be consistent and transparent to avoid creating incentives for gaming appeals.

An independent body, such as the Australian Council for Competition Policy (see Recommendation 43) should be tasked with reporting on the progress of state and territory governments in assessing planning and zoning rules against the public interest test.

For further detail on planning and zoning, see Section 10.1.

## Taxis and ride-sharing

Reform of taxi regulation in most jurisdictions is long overdue. Regulation limiting the number of taxi licences and preventing other services from competing with taxis has raised costs for consumers, including elderly and disadvantaged consumers, and hindered the emergence of innovative passenger transport services. Regulation of taxi and hire car services should be focused on ensuring minimum standards for the benefit of consumers rather than on restricting competition or supporting a particular business model. An independent body should oversee the regulations.

Taxi regulation should be reviewed taking competition into account. Those jurisdictions that have undertaken or are undertaking reviews should implement the reforms.

For further detail on taxi regulation, see Section 10.2.

## **Product standards**

Given that product standards (requirements that goods have certain characteristics) can raise barriers to entry, especially where they are referenced in law (either directly or indirectly) and mandate particular technologies or systems rather than performance outcomes, it is appropriate that they be subject to review. Standards that are not mandated by government should also be reviewed periodically to ensure they do not restrict competition unnecessarily. For example, an Australian Standard that differs unnecessarily from an international standard could limit import competition.

For further detail on product standards, see Section 10.3.

## Recommendation 10 — Priorities for regulation review

Further to Recommendation 8, and in addition to reviewing planning and zoning rules (Recommendation 9), the following should be priority areas for review:

- Taxis and ride-sharing: in particular, regulations that restrict numbers of taxi licences and competition in the taxi industry, including from ride-sharing and other passenger transport services that compete with taxis.
- **Mandatory product standards**: i.e., standards that are directly or indirectly mandated by law, including where international standards can be adopted in Australia.

#### Recommendation 11 — Standards review

Given the unique position of Australian Standards under paragraph 51(2)(c) of the CCA, Australian Standards that are not mandated by government should be subject to periodic review against the public interest test (see Recommendation 8) by Standards Australia.

#### Areas for immediate reform

The Panel identifies the following areas for immediate reform, noting that each area was also identified and reviewed through the NCP process:

- restrictions on retail trading hours (see Recommendation 12);
- parallel import restrictions (see Recommendation 13); and
- pharmacy ownership and location rules (see Recommendation 14).

#### **Retail trading hours**

State and territory governments have deregulated retail trading hours to varying degrees over recent years. This has generally widened choices for consumers. Yet consumers continue to seek greater diversity in how and when they shop, as seen in the rapid take-up of online shopping.

The growing use of the internet for retail purchases is undermining the original intent of restrictions on retail trading hours. When consumers can switch to online suppliers outside regulated trading hours, restrictions on retail trading hours merely serve to disadvantage 'bricks and mortar' retailers relative to their online competitors.

In any event, as more bricks and mortar stores opt for an online presence to counter this disadvantage, the notion of restricted trading hours becomes less meaningful. Customers are already deciding when and how they wish to make purchases. Retailers should be given freedom to respond by deciding for themselves when to open and close their bricks and mortar stores, referring after-hours customers to their online portals.

Regulation of retail trading hours varies across Australia. The Australian Capital Territory, Northern Territory, Victoria, Tasmania and New South Wales have almost completely deregulated retail trading hours, whereas Western Australia, South Australia and Queensland retain significant restrictions.

The Panel believes that deregulation of retail trading hours is overdue, and that remaining restrictions should be removed as soon as possible. To the extent that jurisdictions choose to retain restrictions, these should be strictly limited to Christmas Day, Good Friday and the morning of ANZAC Day. Any public holiday trading restrictions should be applied as broadly as possible to avoid discriminating among different types of retailers.

The Panel notes that a general policy of deregulating trading hours should not prevent jurisdictions from imposing specific restrictions on trading times for alcohol retailing or for gambling services in order to achieve the policy objective of harm minimisation. As noted in Section 10.4, it is certainly not the Panel's view that promoting competition should always trump other legitimate public policy considerations. Instead, regulatory restrictions should be subject to a public interest test to ensure that the benefits of the restrictions to the community as a whole outweigh the costs, and the objectives of the regulation can only be achieved by restricting competition.

The Panel emphasises that deregulation of trading hours does not mean that retailers are obliged to trade 24 hours a day, seven days a week or that all retailers will adopt identical trading hours. Rather, deregulation allows retailers to decide for themselves when to open for trade (with the limited exceptions noted above), as is currently the case in those jurisdictions where retail trading hours are already deregulated. In making this decision, retailers will take into account customer demand and other factors such as labour costs, and requirements of tenancy agreements.

## Recommendation 12 — Retail trading hours

Remaining restrictions on retail trading hours should be removed. To the extent that jurisdictions choose to retain restrictions, these should be strictly limited to Christmas Day, Good Friday and the morning of ANZAC Day, and should be applied broadly to avoid discriminating among different types of retailers. Deregulating trading hours should not prevent jurisdictions from imposing specific restrictions on trading times for alcohol retailing or gambling services in order to achieve the policy objective of harm minimisation.

For further detail on retail trading hours, see Section 10.5.

## **Parallel import restrictions**

Parallel import restrictions are similar to other import restrictions (such as tariffs) in that they benefit local producers by shielding them from international competition. They are an implicit tax on Australian consumers and businesses.

The impact of changing technology and shifting consumer purchasing practices (such as purchasing books online) means that some of these restrictions are easily circumvented. However, removing remaining parallel import restrictions would promote competition and potentially lower prices for consumers.

Many of the concerns raised in submissions around relaxing parallel import restrictions, including concerns about consumer safety, counterfeit products and inadequate enforcement, could be addressed directly through regulation and information. The threat of parallel imports may also induce international suppliers to re-think their regional arrangements.

Relaxing parallel import restrictions should deliver net benefits to the community, provided appropriate regulatory and compliance frameworks and consumer education programs are in place. Transitional arrangements should be considered to ensure that affected individuals and businesses are given adequate notice in advance.

#### Recommendation 13 — Parallel imports

Restrictions on parallel imports should be removed unless it can be shown that:

- the benefits of the restrictions to the community as a whole outweigh the costs; and
- the objectives of the restrictions can only be achieved by restricting competition.

Consistent with the recommendations of recent Productivity Commission reviews, parallel import restrictions on books and second-hand cars should be removed, subject to transitional arrangements as recommended by the Productivity Commission.

Remaining provisions of the *Copyright Act 1968* that restrict parallel imports, and the parallel importation defence under the *Trade Marks Act 1995*, should be reviewed by an independent body, such as the Productivity Commission.

For further detail on parallel imports, see Section 10.6.

## **Pharmacy**

Some ongoing regulation of pharmacy is justified to: uphold patient and community safety; ensure pharmacists provide consumers with appropriate information and advice about their medication; provide equitable access to medication, regardless of a patient's wealth or location; and manage costs to patients and government.

The Australian Government's National Medicines Policy establishes objectives against which medicines are provided and regulations set. The current anti-competitive regulations on the location of pharmacies, or the requirement (with limited exceptions) that only pharmacists own pharmacies, do not appear to serve the objectives of the National Medicines Policy, including the quality of advice provided to consumers. Such restrictions limit both consumers' ability to choose where to obtain pharmacy services and suppliers' ability to meet consumers' demands.

Governments do not need anti-competitive regulation to ensure pharmacies meet community expectations of safety, access and standard of care. A range of alternatives is available, including:

- imposing obligations directly on pharmacies as a condition of their licensing and/or remuneration;
- tendering for the provision of pharmacy services in certain rural or remote areas; or
- a community service obligation, as currently applies to pharmacy wholesaling.

The Panel accepts that competition between pharmacies is not sufficient on its own to meet the access objectives of the National Medicines Policy in rural and remote areas of Australia. The supply of medicines in remote areas is already partly conducted through channels other than retail pharmacies, including through Aboriginal Health Services. That is unlikely to change, even if the current pharmacy location and ownership rules are reformed.

To secure access to medicines for all Australians, governments should consider tendering for the provision of pharmacy services in underserved locations and/or funding through a community service obligation. Since access to medicines is less likely to be an issue in urban settings, the rules targeted at pharmacies in urban areas should continue to be eased at the same time that mechanisms are established to address specific issues concerning access to pharmacies in rural locations.

The Panel recognises that changes to pharmacy location and ownership rules will have a significant impact on the pharmacy sector and that a transition period will therefore be necessary. The Panel

also notes that the current Fifth Community Pharmacy Agreement expires on 1 July 2015, and negotiations for the next agreement are underway. These negotiations provide an opportunity for the Australian Government to implement a further targeted relaxation of the pharmacy location rules as part of a transition towards their eventual removal.

Negotiations will be well underway when this Report is delivered. If changes during the initial years of the new agreement prove too precipitate, there should be provision for a mid-term review to incorporate easing of the location rules later in the life of the Fifth Agreement.

The Panel notes that the recent National Commission of Audit also recommended 'opening up the pharmacy sector to competition, including through the deregulation of ownership and location rules'.<sup>21</sup>

#### Recommendation 14 — Pharmacy

The Panel considers that current restrictions on ownership and location of pharmacies are not needed to ensure the quality of advice and care provided to patients. Such restrictions limit the ability of consumers to choose where to obtain pharmacy products and services, and the ability of providers to meet consumers' preferences.

The Panel considers that the pharmacy ownership and location rules should be removed in the long-term interests of consumers. They should be replaced with regulations to ensure access to medicines and quality of advice regarding their use that do not unduly restrict competition.

Negotiations on the next Community Pharmacy Agreement offer an opportunity for the Australian Government to implement a further targeted relaxation of the location rules, as part of a transition towards their eventual removal. If changes during the initial years of the new agreement prove too precipitate, there should be provision for a mid-term review to incorporate easing of the location rules later in the life of the next Community Pharmacy Agreement.

A range of alternative mechanisms exist to secure access to medicines for all Australians that are less restrictive of competition among pharmacy services providers. In particular, tendering for the provision of pharmacy services in underserved locations and/or funding through a community service obligation should be considered. The rules targeted at pharmacies in urban areas should continue to be eased at the same time that alternative mechanisms are established to address specific issues concerning access to pharmacies in rural locations.

For further detail on pharmacy, see Section 10.7.

## 2.7 COMPETITIVE NEUTRALITY

Stakeholders overwhelmingly support the principle of competitive neutrality and call for Australian governments to recommit to competitive neutrality policy. The Organisation for Economic Co-operation and Development (OECD) also recently stated that, among member nations, Australia has the most complete competitive neutrality framework, backed by separate implementation and complaint-handling mechanisms.

<sup>21</sup> Australian Government 2014, *Towards Responsible Government - The Report of the National Commission of Audit Phase One*, Canberra, page xlii.

But implementing competitive neutrality remains an area of concern for many stakeholders, including small business. The Review's Terms of Reference also direct it to consider the proper boundaries for government in economic activity.

The Panel considers that competitive neutrality policies should be reviewed and updated. Clearer guidelines should be provided on the application of competitive neutrality policies during the start-up stages of government businesses and the period of time over which start-up government businesses should earn a commercial rate of return. The tests used to identify significant business activities should also be reviewed.

Transparency could also be improved by requiring government businesses to report publicly on compliance with competitive neutrality policy and governments to respond publicly to the findings of complaint investigations.

Since each jurisdiction is able to adopt its own approach to competitive neutrality, cross-jurisdiction comparisons can be used to determine 'best practice' as a basis for updating policies and improving current arrangements.

Competitive neutrality policies benefit consumers in markets where both governments and other providers deliver services. This will be especially important in areas where competition policy has yet to reach, such as human services. In these areas, getting the right competitive neutrality policy settings in place will be crucial to securing the benefits of a diverse range of innovative providers. Again, cross-jurisdiction comparisons will help to assess the best ways of achieving competitive neutrality in human services markets. Such feedback could be incorporated into guidelines and practices.

## Recommendation 15 — Competitive neutrality policy

All Australian governments should review their competitive neutrality policies. Specific matters to be considered should include: guidelines on the application of competitive neutrality policy during the start-up stages of government businesses; the period of time over which start-up government businesses should earn a commercial rate of return; and threshold tests for identifying significant business activities.

The review of competitive neutrality policies should be overseen by an independent body, such as the proposed Australian Council for Competition Policy (see Recommendation 43).

## Recommendation 16 — Competitive neutrality complaints

All Australian governments should increase the transparency and effectiveness of their competitive neutrality complaints processes. This should include at a minimum:

- assigning responsibility for investigation of complaints to a body independent of government;
- a requirement for government to respond publicly to the findings of complaint investigations;
   and
- annual reporting by the independent complaints bodies to the proposed Australian Council for Competition Policy (see Recommendation 43) on the number of complaints received and investigations undertaken.

#### Recommendation 17 — Competitive neutrality reporting

To strengthen accountability and transparency, all Australian governments should require government businesses to include a statement on compliance with competitive neutrality principles in their annual reports.

The proposed Australian Council for Competition Policy (see Recommendation 43) should report on the experiences and lessons learned from the different jurisdictions when applying competitive neutrality policy to human services markets.

For further detail on competitive neutrality, see Chapter 13.

## 2.8 GOVERNMENT PROCUREMENT AND OTHER COMMERCIAL ARRANGEMENTS

Government procurement guidelines and decisions can affect the range of goods and services ultimately available to consumers. Procurement can also shape the structure and functioning of competition in markets.

Tender documents have traditionally been written in a prescriptive fashion and with an overarching focus on value for money. Although risk management and value for money are both important considerations, too narrow a focus on these factors can constrain choice, innovation and responsiveness in government-commissioned provision of goods and services.

Tendering with a focus on outcomes, rather than outputs, and trials of less prescriptive tender documents could encourage bidders to suggest new and innovative methods for achieving a government's desired result. Education and information sessions can also help a broader range of businesses understand the procurement process.

Competition principles, particularly those promoting choice and a diversity of providers, should be incorporated into procurement, commissioning, PPP and privatisation policies and practices. Procurement and privatisation policies and practices should also be subject to a public interest test, such that policies and practices should not restrict competition unless the benefits of the restrictions to the community as a whole outweigh the costs, and the objectives of the policy can only be achieved by restricting competition.

### Recommendation 18 — Government procurement and other commercial arrangements

All Australian governments should review their policies governing commercial arrangements with the private sector and non-government organisations, including procurement policies, commissioning, public-private partnerships and privatisation guidelines and processes.

Procurement and privatisation policies and practices should not restrict competition unless:

- the benefits of the restrictions to the community as a whole outweigh the costs; and
- the objectives of the policy can only be achieved by restricting competition.

An independent body, such as the Australian Council for Competition Policy (see Recommendation 43), should be tasked with reporting on progress in reviewing government commercial policies and ensuring privatisation and other commercial processes incorporate competition principles.

For further detail on government procurement and commercial arrangements, see Chapter 14.

## 2.9 ELECTRICITY AND GAS

The Panel acknowledges significant progress in the reform of Australia's electricity and gas sectors. However, reforms have not been finalised and the benefits are yet to be fully realised.

Competition reforms in energy have been a success but have slowed. In Victoria and Queensland, the National Energy Retail Law has yet to be applied without major derogations, undermining the benefits of a national law. Continuing regulation of retail energy prices by jurisdictions other than South Australia, Victoria and New South Wales (though it continues to regulate retail gas prices) perpetuates the distortion of price signals and compromises timely investment in energy infrastructure. The Panel notes that the Queensland Parliament recently legislated to deregulate electricity prices in South East Queensland from 1 July 2015.

The Panel strongly supports moves towards including the Northern Territory and Western Australia into the National Electricity Market, noting that no physical connection is required to do so.

The Panel also supports a detailed review of competition in the gas sector, echoing the proposal within the Eastern Australian Domestic Gas Study, and encourages the Australian Government to commit to undertake such a review through the Energy White Paper.

#### Recommendation 19 — Electricity and gas

State and territory governments should finalise the energy reform agenda, including through:

- application of the National Energy Retail Law with minimal derogation by all National Electricity Market jurisdictions;
- deregulation of both electricity and gas retail prices; and
- the transfer of responsibility for reliability standards to a national framework administered by the proposed Access and Pricing Regulator (see Recommendation 50) and the Australian Energy Market Commission (AEMC).

The Panel supports moves to include Western Australia and the Northern Territory in the National Electricity Market, noting that this does not require physical connection.

The Australian Government should undertake a detailed review of competition in the gas sector.

For further detail on electricity and gas, see Section 11.1.

#### 2.10 WATER

Reform of water has been slower than reform in other sectors. A more national approach to water reform may re-establish its momentum.

If States and Territories implement the principles contained in the National Water Initiative, this will help to re-build momentum in water reform in both the rural and urban sectors. The Panel notes that, in general, urban water pricing fails to reflect its cost of provision and this is discouraging private sector participation in providing urban water.

The National Water Initiative outlines principles for best-practice pricing of urban water and the Panel sees benefit in the ACCP working with state and territory regulators to assist jurisdictions in applying those principles, allowing for necessary jurisdictional differences. Following this, the ACCP

should work with all States and Territories to develop plans for fully implementing the National Water Initiative.

The Panel expects that, should any of the regulatory functions in water markets be transferred to a national framework, the national aspects would be administered by the proposed Access and Pricing Regulator (APR). Notwithstanding, States and Territories should retain the option to transfer national regulation to the APR or to a suitably accredited state regulator.

#### Recommendation 20 — Water

All governments should progress implementation of the principles of the National Water Initiative, with a view to national consistency. Governments should focus on strengthening economic regulation in urban water and creating incentives for increased private participation in the sector through improved pricing practices.

State and territory regulators should collectively develop best-practice pricing guidelines for urban water, with the capacity to reflect necessary jurisdictional differences. To ensure consistency, the Australian Council for Competition Policy (see Recommendation 43) should oversee this work.

State and territory governments should develop clear timelines for fully implementing the National Water Initiative, once pricing guidelines are developed. The Australian Council for Competition Policy should assist States and Territories to do so.

Where water regulation is made national, the responsible body should be the proposed national Access and Pricing Regulator (see Recommendation 50) or a suitably accredited state body.

For further detail on water, see Section 11.2.

### 2.11 INFORMED CHOICE

Globalisation, competition and technological innovation have expanded the range of businesses from which Australian consumers can choose to purchase goods and services. The Panel is also recommending that user choice be placed at the heart of human services delivery, and that governments further their efforts to encourage a diversity of providers.

Greater choice can act as a powerful force to drive innovation in markets for goods and services, but it also means that consumers need to know more about markets if they are to secure the best deals.

The Panel believes that markets work best when consumers are informed and engaged, empowering them to make good decisions. Empowering consumers requires that they have access to accurate, easily understood information about products and services on offer.

But just providing information is not enough to guarantee good choices by consumers. The 'right' type of information must also be provided, so consumers can (and want to) act on the available information. Insights from psychology and behavioural economics suggest that consumers can have behavioural traits that prevent them from making good use of even well-presented information. Governments should take account of these findings to ensure that consumers are able to enjoy the full benefits of competition and choice.

Businesses are collecting more and more data, notably through customer loyalty cards, to better understand their customers. The Panel sees scope for Australian consumers' access to data to be improved to better inform their decisions.

## Recommendation 21 — Informed choice

Governments should work with industry, consumer groups and privacy experts to allow consumers to access information in an efficient format to improve informed consumer choice.

The proposed Australian Council for Competition Policy (see Recommendation 43) should establish a working group to develop a partnership agreement that both allows people to access and use their own data for their own purposes and enables new markets for personal information services. This partnership should draw on the lessons learned from similar initiatives in the US and UK.

Further, governments, both in their own dealings with consumers and in any regulation of the information that businesses must provide to consumers, should draw on lessons from behavioural economics to present information and choices in ways that allow consumers to access, assess and act on them.

For further detail on informed choice, see Chapter 16.

## 3 COMPETITION LAW

## 3.1 SIMPLIFICATION

The Panel has asked the following questions in guiding its consideration of whether the *Competition* and *Consumer Act 2010* (CCA) is fit for purpose:

- Does the law focus on enhancing consumer wellbeing over the long term?
- Does the law protect competition rather than protecting individual competitors?
- Does the law strike the right balance between prohibiting anti-competitive conduct and not interfering with efficiency, innovation and entrepreneurship?
- Is the law as clear, simple and predictable as it can be?

The Panel supports the general form and structure of the CCA, that is:

- The law prohibits specific categories of anti-competitive conduct, with economy-wide application.
- Only conduct that is anti-competitive in most circumstances is prohibited per se other
  conduct is prohibited only if it has the purpose, effect or likely effect of substantially lessening
  competition.
- Contraventions of the law are adjudicated by a court, with proceedings able to be initiated by a public regulatory authority or through private suit.
- Business can seek exemption from the law in individual cases on public benefit grounds.

## Recommendation 22 — Competition law concepts

The central concepts, prohibitions and structure enshrined in the current competition law should be retained, since they are appropriate to serve the current and projected needs of the Australian economy.

However, the Panel considers that the competition law provisions of the CCA, including the provisions regulating the granting of exemptions, are unnecessarily complex.

Complex law imposes costs on the economy: direct costs caused by the need for legal advice and prolonged legal disputation; and indirect costs caused by business and regulatory uncertainty.

The competition law provisions of the CCA would benefit from simplification, while retaining their underlying policy intent.

#### Recommendation 23 — Competition law simplification

The competition law provisions of the CCA should be simplified, including by removing overly specified provisions and redundant provisions.

The process of simplifying the CCA should involve public consultation.

Provisions that should be removed include:

- subsection 45(1) concerning contracts made before 1977; and
- sections 45B and 45C concerning covenants.

For further detail on competition law concepts, see Chapter 17.

## 3.2 Application to government activities in trade or commerce

As a consequence of the Hilmer Review, the CCA was extended to apply to the Crown, but only insofar as the Crown carries on a business, either directly or through an authority of the Crown.

There are many circumstances in which the Crown (whether as a department or an authority) undertakes commercial transactions but does not carry on a business. This is particularly the case in procurement, whether for delivering large infrastructure projects or the regular requirements of the health or education systems.

Through commercial transactions entered into with market participants, the Crown (whether in right of the Commonwealth, state, territory or local governments) has the potential to harm competition (see Recommendation 18). The Panel considers that the Hilmer reforms should be carried a step further, with the Crown subject to the competition law insofar as it undertakes activity in trade or commerce.

## **Recommendation 24** — Application of the law to government activities

Sections 2A, 2B and 2BA of the CCA should be amended so that the competition law provisions apply to the Crown in right of the Commonwealth and the States and Territories (including local government) insofar as they undertake activity in trade or commerce.

This recommendation is reflected in the model legislative provisions in Appendix A.

For further detail on the application of competition laws to government activities, see Section 14.2.

## 3.3 MARKET DEFINITION

The Panel considers that the competition law provisions of the CCA are correctly focused on conduct that damages competition in markets in Australia and that the current definition of 'market' (being a market in Australia) is appropriate.

This reflects the object of the law to protect the welfare of Australians. There is no sound reason for Australian law to regulate conduct affecting competition in overseas markets.

However, this should not mean that the CCA ignores forces of competition arising outside Australia but which affect Australian markets. Frequently, the sources of competition in Australian markets

originate globally, especially as increasing numbers of Australian consumers purchase goods and services online from overseas suppliers.

The definition of the term 'competition' in the CCA is important. The CCA has been framed to take account of all sources of competition that affect markets in Australia, with the term defined to include competition from imported goods and services.

Nevertheless, given the importance of ensuring that global sources of competition are considered where relevant, the current definition of 'competition' in the CCA should be strengthened so that there can be no doubt that it includes competition from *potential* imports of goods and services, not just *actual* imports.

The Panel does not intend that this change would expand market definitions in the competition law to include every product and service that could conceivably be imported into Australia, but only clarify that the credible threat of import competition is a relevant component of a competition analysis.

#### Recommendation 25 — Definition of market and competition

The current definition of 'market' in section 4E of the CCA should be retained but the current definition of 'competition' in section 4 should be amended to ensure that competition in Australian markets includes competition from goods imported or capable of being imported, or from services rendered or capable of being rendered, by persons not resident or not carrying on business in Australia.

This recommendation is reflected in the model legislative provisions in Appendix A.

For further detail on market definition, see Section 18.1.

## 3.4 EXTRA-TERRITORIAL REACH OF THE LAW

The Panel considers that the competition law provisions of the CCA ought to apply to firms engaging in conduct outside Australia if that conduct relates to trade or commerce within Australia or between Australia and places outside Australia. The application of the law in those circumstances ought not to depend on whether the firm is incorporated in, or carries on business within, Australia.

Private actions are also an important part of the competition law framework. The requirement for private parties to seek ministerial consent in connection with proceedings involving conduct that occurs outside Australia is an unnecessary roadblock to possible redress for harm suffered as a result of a breach of Australian competition law.

#### Recommendation 26 — Extra-territorial reach of the law

Section 5 of the CCA, which applies the competition law to certain conduct engaged in outside Australia, should be amended to remove the requirement that the contravening firm has a connection with Australia in the nature of residence, incorporation or business presence and to remove the requirement for private parties to seek ministerial consent before relying on extra-territorial conduct in private competition law actions. Instead, the competition law should apply to overseas conduct insofar as the conduct relates to trade or commerce within Australia or between Australia and places outside Australia.

The in-principle view of the Panel is that the foregoing changes should also be made in respect of actions brought under the Australian Consumer Law.

This recommendation is reflected in the model legislative provisions in Appendix A.

For further detail on extra-territorial reach of the law, see Section 23.2.

## 3.5 CARTELS

Cartel conduct between competitors is anti-competitive in most circumstances and should be prohibited per se. The Panel supports the intent of the cartel conduct prohibitions, including the combined criminal and civil sanctions.

However, the Panel sees significant deficiencies in the current framework of the cartel prohibitions, particularly having regard to its criminal sanctions. Specifically, the Panel considers that:

- The provisions are excessively complex, which undermines compliance and enforcement.
- The cartel provisions, consistent with Australia's competition laws generally, should be confined to cartel conduct involving persons who compete to supply goods or services to, or acquire goods or services from, persons resident in or carrying on a business within Australia.
- Given the potential for criminal sanctions, the provisions ought to be confined to conduct involving firms that are actual competitors and not firms for whom competition is a mere possibility.
- Joint ventures and similar forms of business collaboration should not be subject to cartel prohibitions and should only be unlawful if they substantially lessen competition.
- Similarly, trading restrictions that are imposed by one firm on another in connection with the supply or acquisition of goods or services (including IP licensing) should not be subject to cartel prohibitions, and should only be unlawful if they substantially lessen competition.

#### Recommendation 27 — Cartel conduct prohibition

The prohibitions against cartel conduct in Part IV, Division 1 of the CCA should be simplified and the following specific changes made:

- The provisions should apply to cartel conduct involving persons who compete to supply goods or services to, or acquire goods or services from, persons resident in or carrying on business within Australia.
- The provisions should be confined to conduct involving firms that are actual or likely competitors, where 'likely' means on the balance of probabilities.
- A broad exemption should be included for joint ventures, whether for the production, supply, acquisition or marketing of goods or services, recognising that such conduct will be prohibited by section 45 of the CCA if it has the purpose, effect or likely effect of substantially lessening competition.
- An exemption should be included for trading restrictions that are imposed by one firm on another in connection with the supply or acquisition of goods or services (including intellectual property licensing), recognising that such conduct will be prohibited by section 45 of the CCA (or section 47 if retained) if it has the purpose, effect or likely effect of substantially lessening competition.

This recommendation is reflected in the model legislative provisions in Appendix A.

The Panel also considers that the per se prohibition of exclusionary provisions, as defined in section 4D, is no longer necessary since, in practice, such conduct is materially the same as cartel conduct in the form of market sharing.

Accordingly, the Panel believes that the prohibition against exclusionary provisions should be removed from the CCA.

## Recommendation 28 — Exclusionary provisions

The CCA should be amended to remove the prohibition of exclusionary provisions in subparagraphs 45(2)(a)(i) and 45(2)(b)(i), with an amendment to the definition of cartel conduct to address any resulting gap in the law.

This recommendation is reflected in the model legislative provisions in Appendix A.

For further detail on cartel conduct, see Section 20.1.

## 3.6 ANTI-COMPETITIVE DISCLOSURE OF INFORMATION

The Panel considers that, in their current form, the prohibitions against 'price signalling' in the CCA do not strike the right balance in distinguishing between anti-competitive and pro-competitive conduct. Being confined in their operation to a single industry (banking), the current provisions are also inconsistent with the principle that the CCA should apply to all businesses generally.

The Panel considers that public price disclosure can help consumers make informed choices and is unlikely to raise significant competition concerns. Accordingly, the Panel believes there is no sound basis for prohibiting public price disclosure, either in the banking industry or more generally.<sup>22</sup>

Private price disclosure to a competitor will generally have more potential to harm competition, as it may be used to facilitate collusion among competitors. However, private disclosure may be necessary under some business circumstances or in the ordinary course of business, particularly in connection with joint ventures or similar types of business collaboration. For that reason, a per se prohibition has the potential to overreach.

The Panel considers that anti-competitive price signalling does not need its own separate Division in the CCA; rather, price signalling can be addressed by extending section 45 to cover concerted practices that have the purpose, effect or likely effect of substantially lessening competition.

The word 'concerted' means jointly arranged or carried out or co-ordinated. Hence, a concerted practice between market participants is a practice that is jointly arranged or carried out or co-ordinated between the participants. The expression 'concerted practice with one or more other persons' conveys that the impugned practice is neither unilateral conduct nor mere parallel conduct by market participants (e.g., suppliers selling products at the same price).

The Panel proposes that such conduct would only be prohibited if it can be shown that the concerted practice has the purpose, effect or likely effect of substantially lessening competition. The Panel considers that the word 'concerted' has a clear and practical meaning and no further definition is required for the purposes of a legal enactment.

The Panel does not consider that the cartel conduct prohibitions should be expanded to include concerted practices. The Panel considers that imposing criminal sanctions for cartel conduct should require proof of a contract, arrangement or understanding between competitors.

## **Recommendation 29 — Price signalling**

The 'price signalling' provisions of Part IV, Division 1A of the CCA are not fit for purpose in their current form and should be repealed.

Section 45 should be extended to prohibit a person engaging in a concerted practice with one or more other persons that has the purpose, effect or likely effect of substantially lessening competition.

This recommendation is reflected in the model legislative provisions in Appendix A.

For further detail on anti-competitive disclosure of information, see Section 20.2.

## 3.7 MISUSE OF MARKET POWER

An effective provision to deal with unilateral anti-competitive conduct is a necessary part of the competition law. This is particularly the case in Australia, where the small size of the Australian

The Panel notes that the prohibition on certain public disclosures also applies to disclosures of a corporation's capacity or commercial strategy. The Report does not deal with these matters separately, since the Panel considers that the same issues arise as in the case of public price disclosure.

economy frequently leads to concentrated markets. The Panel considers that section 46 can be re-framed in a manner that will improve its effectiveness in targeting anti-competitive unilateral conduct and focus it more clearly on the long-term interests of consumers.

The Panel regards the threshold test of 'substantial degree of power in a market' as appropriate and well understood. In contrast, the 'take advantage' limb of section 46 is not a useful test by which to distinguish competitive from anti-competitive unilateral conduct. This test has given rise to substantial difficulties of interpretation, which have been revealed in the decided cases, undermining confidence in the effectiveness of the law.

Perhaps more significantly, the test is not best adapted to identifying a misuse of market power. Business conduct should not be immunised merely because it is often undertaken by firms without market power. Conduct such as exclusive dealing, loss-leader pricing and cross-subsidisation may all be undertaken by firms without market power without raising competition concerns, while the same conduct undertaken by a firm with market power might raise competition concerns.

Further, the focus of the prohibition on showing the purpose of damaging a competitor is inconsistent with the overriding policy objective of the CCA to protect competition, and not individual competitors. The prohibition ought to be directed to conduct that has the purpose or effect of harming the competitive process.

The Panel also considers that the supplementary prohibitions, which attempt to address concerns about predatory pricing, <sup>23</sup> do not advance the policy intent of section 46.

Accordingly, the Panel proposes that the primary prohibition in section 46 be re-framed to prohibit a corporation that has a substantial degree of power in a market from engaging in conduct if the conduct has the purpose, effect or likely effect of substantially lessening competition in that or any other market. The proposed test of 'substantial lessening of competition' is the same test as is found in section 45 (anti-competitive agreements), section 47 (exclusive dealing) and section 50 (mergers) of the CCA, and the test is well accepted within those sections.

Conduct undertaken by a firm with substantial market power can have pro-competitive and anti-competitive features. The issue for courts, and for firms assessing their own conduct, is to weigh the pro-competitive and anti-competitive impacts of the conduct to decide if there has been a substantial lessening of competition. To clarify the law and mitigate concerns about over-capture, the Panel proposes that section 46 include legislative guidance with respect to the intended operation of the section. Specifically, the legislation should direct the court, when determining whether conduct has the purpose, effect or likely effect of substantially lessening competition in a market, to have regard to the extent to which the conduct:

- increases competition in a market, including by enhancing efficiency, innovation, product quality or price competitiveness; and
- lessens competition in a market, including by preventing, restricting or deterring the potential for competitive conduct in a market or new entry into a market.

The proposed reform to section 46 is intended to improve its clarity, force and effectiveness so that it can be used to prevent unilateral conduct that substantially harms competition and that has no economic justification.

<sup>23</sup> Competition and Consumer Act 2010 subsections 46(1AAA) and (1AA).

#### Recommendation 30 — Misuse of market power

The primary prohibition in section 46 of the CCA should be re-framed to prohibit a corporation that has a substantial degree of power in a market from engaging in conduct if the proposed conduct has the purpose, or would have or be likely to have the effect, of substantially lessening competition in that or any other market.

To mitigate concerns about inadvertently capturing pro-competitive conduct, the legislation should direct the court, when determining whether conduct has the purpose, effect or likely effect, of substantially lessening competition in a market, to have regard to:

- the extent to which the conduct has the purpose, effect or likely effect of increasing competition in the market, including by enhancing efficiency, innovation, product quality or price competitiveness; and
- the extent to which the conduct has the purpose, effect or likely effect of lessening competition in the market, including by preventing, restricting or deterring the potential for competitive conduct in the market or new entry into the market.

Such a re-framing would allow the provision to be simplified. Amendments introduced since 2007 would be unnecessary and could be repealed. These include specific provisions prohibiting predatory pricing, and amendments clarifying the meaning of 'take advantage' and how the causal link between the substantial degree of market power and anti-competitive purpose may be determined.

Authorisation should be available in relation to section 46, and the ACCC should issue guidelines regarding its approach to the provision.

This recommendation is reflected in the model legislative provisions in Appendix A.

For further detail on misuse of market power, see Section 19.1.

#### 3.8 Unconscionable conduct

Both the business and the wider community expect business to be conducted according to a minimum standard of fair dealing. There are sound economic and social reasons for enshrining minimum standards of fair dealing within the law.

The Panel has heard concerns expressed by small businesses and suppliers in respect of behaviours of larger businesses in their supply chains. The business unconscionable conduct provisions were introduced specifically to address these concerns.

Enforcing business-to-business unconscionable conduct provisions is an important function of the Australian Competition and Consumer Commission (ACCC). The Panel notes the recent Federal Court declarations in two proceedings instituted by the ACCC that Coles engaged in unconscionable conduct in 2011 in its dealings with certain suppliers. These cases indicate that the current unconscionable conduct provisions are working as intended to meet their policy goals.

Active and ongoing review of these provisions should occur as other matters arise. If deficiencies in the operation of the provisions become evident, they should be remedied promptly.

For further detail on unconscionable conduct, see Section 19.4.

## 3.9 PRICE DISCRIMINATION

The Panel recognises that some small businesses and consumers have concerns about the impacts of price discrimination. However, the former prohibition on price discrimination (contained in the former section 49 of the CCA) was found likely to result in price inflexibility, which would undermine consumer welfare.

In relation to international price discrimination, the Panel considers that any attempt to prohibit this would face significant implementation difficulties. A prohibition on international price discrimination could also have significant negative consequences, ultimately limiting consumer choice. Instead, the Panel favours encouraging the development and use of lawful market-based mechanisms to put downward pressure on prices.

#### Recommendation 31 — Price discrimination

A specific prohibition on price discrimination should not be reintroduced into the CCA. Where price discrimination has an anti-competitive impact on markets, it can be dealt with by the existing provisions of the law (including through the Panel's recommended revisions to section 46 (see Recommendation 30)).

Attempts to prohibit international price discrimination should not be introduced into the CCA on account of significant implementation and enforcement complexities and the risk of negative unintended consequences. Instead, the Panel supports moves to address international price discrimination through market solutions that empower consumers. These include removing restrictions on parallel imports (see Recommendation 13) and ensuring that consumers are able to take lawful steps to circumvent attempts to prevent their access to cheaper legitimate goods.

For further detail on price discrimination, see Section 19.3.

# 3.10 Vertical restrictions (other than resale price maintenance)

As a general principle, the Panel believes that the CCA should not interfere with trading conditions agreed between buyers and sellers in connection with acquiring and supplying goods and services, unless those conditions have the purpose, effect or likely effect of substantially lessening competition.

Section 47 prohibits most vertical restrictions only if they have the purpose, effect or likely effect of substantially lessening competition. The one exception is third-line forcing. Under the CCA, third-line forcing is prohibited per se — that is, regardless of the purpose or effect of the conduct.

The Panel sees no need for third-line forcing to be singled out from other forms of vertical trading conditions and be prohibited per se. As notifications to the ACCC demonstrate, third-line forcing is a common business practice and rarely has anti-competitive effects.

#### Recommendation 32— Third-line forcing test

Third-line forcing (subsections 47(6) and (7) of the CCA) should only be prohibited where it has the purpose, effect or likely effect of substantially lessening competition.

The Panel agrees with the view expressed in many submissions that section 47 is unnecessarily complex and therefore difficult for business to understand and apply. The section focuses attention

on particular forms of vertical restraints and directs attention away from the central issue — whether the restriction is anti-competitive.

The amendments to section 46 recommended in this Report (see Recommendation 30) will render section 47 redundant. Section 45 will apply to all vertical restraints (including third-line forcing) included in a contract, arrangement or understanding; section 46 will apply if a corporation refuses to supply goods or services because the acquirer will not agree to accept a vertical restraint (including third-line forcing). As amended, section 45 and section 46 would apply the same competition test as in section 47.

Section 46 has an additional limitation not expressed in section 47: the prohibition only applies to a corporation that has substantial market power. However, this will not limit the effectiveness of the law. It is well accepted that vertical restrictions will not substantially lessen competition unless they are imposed by a corporation with substantial market power.

The Panel therefore recommends that section 47 be repealed, simplifying the competition law.

If section 46 is not amended as recommended, the Panel considers that section 47 should be simplified along the lines proposed in the Draft Report. The Panel has included a simplified form of section 47 in the model legislative provisions in Appendix A. The model form takes account of submissions received in response to the Draft Report.

#### Recommendation 33 — Exclusive dealing coverage

Section 47 of the CCA should be repealed and vertical restrictions (including third-line forcing) and associated refusals to supply addressed by sections 45 and 46 (as amended in accordance with Recommendation 30).

For further detail on vertical restrictions (other than resale price maintenance), see Section 20.3.

## 3.11 RESALE PRICE MAINTENANCE

The appropriateness of a per se prohibition on resale price maintenance (RPM) has been debated for many years, both in Australia and overseas. When the per se prohibition was enacted in Australia in the mid-1970s, it reflected the law in many comparable jurisdictions. However, over the last 20 years some countries — particularly the US and Canada — have moved away from the per se prohibition of RPM. Other jurisdictions, including Europe and New Zealand, have retained the per se prohibition.

At this time, the Panel sees no sufficient case for changing the prohibition of RPM from a per se prohibition to a competition-based test. However, it would be appropriate to allow business to seek exemption from the prohibition more easily. This could be achieved through allowing RPM to be assessed through the notification process, which is quicker and less expensive for businesses than authorisation. This change would also have the advantage of allowing the ACCC to assess RPM trading strategies more frequently, and thereby provide better evidence as to the competitive effects of RPM in Australia.

A general tenet of the competition law is that companies within a corporate group are treated as a single economic entity and are not considered to be competitors. For that reason, the prohibitions in

sections 45 and 47 do not apply to trading arrangements entered into between related companies.<sup>24</sup> A similar principle ought to apply to RPM. Currently, there is no exemption for RPM between a manufacturer and a retailer that is a subsidiary of the manufacturer.

# Recommendation 34 — Resale price maintenance

The prohibition on resale price maintenance (RPM) in section 48 of the CCA should be retained in its current form as a per se prohibition, but notification should be available for RPM conduct.

This recommendation is reflected in the model legislative provisions in Appendix A.

The prohibition should also be amended to include an exemption for RPM conduct between related bodies corporate, as is the case under sections 45 and 47.

For further detail on resale price maintenance, see Section 20.4.

## 3.12 MERGERS

The Panel considers that the current prohibition of mergers that are likely to substantially lessen competition in Australian markets is appropriate.

Concerns have been raised that Australia's merger law does not give proper consideration to global markets within which many businesses compete. Some submissions argue that the term 'market' in the CCA is defined as a market 'in Australia' and that causes the competition analysis to be narrowly focused. As noted above (see Section 3.3), although the Panel considers that the CCA correctly focuses upon conduct that damages competition in markets in Australia (to protect Australian consumers), the CCA has been framed to take account of all sources of competition that affect Australian markets. Recommendation 25 is intended to strengthen that principle.

Although some submissions raise concerns that the ACCC opposes too few mergers, others question whether the ACCC's application of the CCA is constraining Australian businesses from achieving efficient scale through mergers to become globally competitive. To compete effectively, businesses must continuously pursue economic efficiency. In many industries, efficiency requires scale. Businesses may pursue mergers to achieve efficient scale to compete more effectively in global markets.

In many markets in Australia, achieving efficient scale will not substantially lessen competition because of the constraining influence of imports. Such mergers are allowed under the CCA. However, in some markets, the opposite will be the case: the influence of imports may be weak and unable to constrain the resulting market power of the merged businesses. When that occurs, conflicting interests emerge: the gain to the businesses that wish to merge through achieving greater efficiency against the potential detriment to Australian consumers arising from reduced competition.

The Panel considers that the CCA has sufficient flexibility to allow such issues to be adjudicated and determined by the ACCC or the Australian Competition Tribunal (the Tribunal). The merger authorisation process applies a public benefit test that covers all potential benefits and detriments, including economies of scale. There may be occasions where it is in the public interest to allow a

<sup>24</sup> Competition and Consumer Act 2010, subsections 45(8) and 47(12).

particular merger to achieve efficient scale to compete globally, notwithstanding that the merger adversely affects competition in Australia.

Nonetheless, the Panel considers that the administration of the merger law can be improved.

There is widespread support for retaining the ACCC's informal merger review process. However, strong concerns have been expressed about the timeliness and transparency of the process.

The Panel considers that it is not sensible to attempt to regulate an informal process which, by definition, operates outside any formal legal framework. The flexibility of the informal process is widely recognised as being beneficial and should not be interfered with. However, the public interest is served by timely merger decisions and by transparency in the public administration of the merger law. The Panel sees scope for further consultation between the ACCC and business representatives, with the objective of developing an informal review process that delivers more timely decisions. The Panel also considers that ex-post evaluations of some merger decisions could be undertaken by the proposed Australian Council for Competition Policy (ACCP) (see Recommendation 44) to draw lessons for future merger reviews (but not to overturn past decisions).

The Panel considers that concerns about the timeliness and transparency of merger review processes can also be addressed through a more streamlined formal exemption process. The current formal exemption processes are excessively complex and prescriptive, being a formal clearance application to the ACCC and an alternative authorisation application to the Tribunal. This has deterred the use of these mechanisms and fuelled complaints about the way the informal process is applied to large mergers that involve contested facts and issues.

The Panel also considers that, if a more streamlined formal exemption process were introduced, it would be preferable for the ACCC to be the first instance decision-maker rather than the Tribunal. Having regard to its composition and powers, the ACCC is better suited to investigation and first-instance decision making in administering the competition law (including mergers), while the Tribunal is better suited to an appellate or review role.

The Tribunal's review of the ACCC's decision should be based upon the material that was before the ACCC, but the Tribunal should have the discretion to allow a party to adduce further evidence, or to call and question a witness, if the Tribunal is satisfied that there is sufficient reason. A full rehearing, with an unfettered ability for parties to put new material before the Tribunal, would be likely to dampen the incentive to put all relevant material to the ACCC in the first instance, and may lead to delays if the Tribunal has to deal with large amounts of new evidence. On the other hand, circumstances may arise in which it is reasonable to allow new evidence to be provided to the Tribunal. Further, the Tribunal may also consider that it would be assisted by hearing directly from witnesses relied on by the ACCC, through questioning by the parties and/or the Tribunal.

# Creeping acquisitions

Concerns about 'creeping acquisitions' typically arise where a business with a substantial degree of market power acquires many small competitors over time.

A legitimate question arises regarding whether, in assessing the likely effect of a proposed merger, the merger provisions of the CCA should also take account of the aggregate effect of the corporation's previous acquisitions within, for example, the prior three years. The complicating factor is that market conditions may have altered materially over the period chosen. Such a change would impose additional costs associated with merger review. On balance, in the absence of evidence of harmful acquisitions proceeding because of a gap in the law on creeping acquisitions, the Panel does not consider that a sufficiently strong case for change has been made.

#### Recommendation 35 — Mergers

There should be further consultation between the ACCC and business representatives with the objective of delivering more timely decisions in the informal merger review process.

The formal merger exemption processes (i.e., the formal merger clearance process and the merger authorisation process) should be combined and reformed to remove unnecessary restrictions and requirements that may have deterred their use. The specific features of the review process should be settled in consultation with business, competition law practitioners and the ACCC.

However, the general framework should contain the following elements:

- The ACCC should be the decision-maker at first instance.
- The ACCC should be empowered to authorise a merger if it is satisfied that the merger does not substantially lessen competition or that the merger would result, or would be likely to result, in a benefit to the public that would outweigh any detriment.
- The formal process should not be subject to any prescriptive information requirements, but the ACCC should be empowered to require the production of business and market information.
- The formal process should be subject to strict timelines that cannot be extended except with the consent of the merger parties.
- Decisions of the ACCC should be subject to review by the Australian Competition Tribunal under a process that is also governed by strict timelines.
- The review by the Australian Competition Tribunal should be based upon the material that was before the ACCC, but the Tribunal should have the discretion to allow a party to adduce further evidence, or to call and question a witness, if the Tribunal is satisfied that there is sufficient reason.

Merger review processes and analysis would also be improved by implementing a program of post-merger evaluations, looking back on a number of past merger decisions to determine whether the ACCC's processes were effective and its assessments borne out by events. This function could be performed by the Australian Council for Competition Policy (see Recommendation 44).

For further detail on mergers, see Chapter 18.

#### 3.13 Secondary boycotts and employment-related matters

The negotiation of employment terms and conditions (remuneration, conditions of employment, hours of work or working conditions of employees) has always been excluded from most of the competition law provisions of the CCA by paragraph 51(2)(a). The reason for this exclusion is that the negotiation and determination of employment terms and conditions is governed by a separate regulatory regime, currently contained in the *Fair Work Act 2009*. The policy rationale is that labour markets are not in all respects comparable to other product or service markets. As a general principle, the Panel agrees with that view.

However, two categories of employment-related conduct do not fall within that general exclusion:

- secondary boycotts, which are prohibited by sections 45D, 45DA and 45DB; and
- trading restrictions in industrial agreements, which are prohibited by sections 45E and 45EA.

# Secondary boycotts

A strong case remains for the CCA to retain the prohibition of secondary boycotts. A sufficient case has not been made to limit the scope of the secondary boycott prohibition, nor to broaden the scope of the exception for employment-related matters.

The Panel did not receive compelling evidence of actual secondary boycott activity falling within the environmental and consumer protection exception in the CCA. In the absence of such evidence, the Panel does not see an immediate case for amending the exception. However, if such evidence arises from future boycott activity, the exceptions should be reassessed.

Some industry organisations, especially in building, construction and mining, believe that public enforcement of the secondary boycott provisions is inadequate, a point emphasised in the Interim Report of the Royal Commission into Trade Union Governance and Corruption. <sup>25</sup> Timely and effective public enforcement serves as a deterrent to boycott activity and needs to exist both in regulatory culture and capability. The Panel believes that the ACCC should pursue secondary boycott cases with increased vigour, comparable to that which it applies in pursuing other contraventions of the competition law.

It would be useful for the ACCC to report the number of complaints it receives about different parts of the CCA, including secondary boycotts, and the manner in which the complaints are resolved.

Further, the Panel sees no reason why the maximum pecuniary penalties for breaches of secondary boycott provisions should be lower than those for other breaches of the competition law.

## Recommendation 36 — Secondary boycotts

The prohibitions on secondary boycotts in sections 45D-45DE of the CCA should be maintained and effectively enforced.

The ACCC should pursue secondary boycott cases with increased vigour, comparable to that which it applies in pursuing other contraventions of the competition law. It should also publish in its annual report the number of complaints made to it in respect of different parts of the CCA, including secondary boycott conduct and the number of such matters investigated and resolved each year.

The maximum penalty level for secondary boycotts should be the same as that applying to other breaches of the competition law.

# Trading restrictions in industrial agreements

Section 45E of the CCA prohibits a person (an employer) from making a contract, arrangement or understanding with an organisation of employees that contains a provision restricting the freedom of the employer to supply goods or services to, or acquire goods or services from, another person. Section 45EA prohibits a person from giving effect to such a contract, arrangement or understanding. The Panel considers that sections 45E and 45EA are important provisions that protect trading freedoms.

<sup>25</sup> Heydon J. D., AC QC 2014, Royal Commission into Trade Union Governance and Corruption Interim Report, Volume 2, page 1106.

There appears to be a possible conflict between the intended operation of sections 45E and 45EA and the regulation of awards and industrial agreements under the Fair Work Act. This issue has been brought into focus by the 2012 decision of the Full Court of the Federal Court, *Australian Industry Group v Fair Work Australia* [2012] FCAFC 108.

It appears to be lawful under the *Fair Work Act 2009* to make awards and register enterprise agreements that place restrictions on the freedom of employers to engage contractors or source certain goods or non-labour services. The Panel considers it desirable that this apparent conflict be resolved.

The Panel favours competition over restrictions and believes that businesses should generally be free to supply and acquire goods and services, including contract labour, should they choose. Accordingly, the Panel considers that sections 45E and 45EA should be amended so that they expressly apply to awards and industrial agreements, except to the extent they deal with the remuneration, conditions of employment, hours of work or working conditions of employees.

The Panel considers that the ACCC should be given the right to intervene in proceedings (i.e., to be notified, appear and be heard subject to time limits) before the Fair Work Commission and make submissions concerning compliance with sections 45E and 45EA. The ACCC and Fair Work Commission should establish a protocol to govern these arrangements.

Further, the present limitation in sections 45E and 45EA, such that the prohibition only applies to restrictions affecting persons with whom an employer 'has been accustomed, or is under an obligation', to deal, should be removed, and the maximum penalty for breaches of these provisions should be in line with those for breaches of the rest of the competition law.

## **Recommendation 37 — Trading restrictions in industrial agreements**

Sections 45E and 45EA of the CCA should be amended so that they apply to awards and industrial agreements, except to the extent they relate to the remuneration, conditions of employment, hours of work or working conditions of employees.

Further, the present limitation in sections 45E and 45EA, such that the prohibitions only apply to restrictions affecting persons with whom an employer 'has been accustomed, or is under an obligation,' to deal, should be removed.

These recommendations are reflected in the model provisions in Appendix A.

The ACCC should be given the right to intervene in proceedings before the Fair Work Commission and make submissions concerning compliance with sections 45E and 45EA. A protocol should be established between the ACCC and the Fair Work Commission.

The maximum penalty for breaches of sections 45E and 45EA should be the same as that applying to other breaches of the competition law.

For further detail on secondary boycotts, and trading restrictions in industrial agreements, see Chapter 21.

## 3.14 EXEMPTION PROCESSES

The exemption processes of authorisation and notification included in the CCA are important. They recognise that, in certain circumstances, particular conduct may not harm competition or may give rise to public benefits that outweigh any competitive harm.

Like much of the CCA, the authorisation and notification procedures have become overly complex, which imposes costs on business. Wherever possible, the Panel supports removing unnecessary complexity.

Significant steps can be taken to simplify the authorisation and notification procedures. First, in respect of authorisation, it should be permissible to apply for authorisation of a business arrangement through a single application and without regard to the specific provisions of the CCA that might be contravened by the proposed conduct.

Second, for both authorisation and notification, the ACCC should be empowered to grant the exemption (other than in respect of the per se prohibitions) if it is satisfied that either the proposed conduct is unlikely to substantially lessen competition or that the proposed conduct is likely to result in a net public benefit.

Each of these changes would assist in focusing the exemption process on the issues of substance and away from technicalities.

#### Recommendation 38 — Authorisation and notification

The authorisation and notification provisions in Part VII of the CCA should be simplified to:

- ensure that only a single authorisation application is required for a single business transaction or arrangement; and
- empower the ACCC to grant an exemption from sections 45, 46 (as proposed to be amended), 47 (if retained) and 50 if it is satisfied that the conduct would not be likely to substantially lessen competition or that the conduct would result, or would be likely to result, in a benefit to the public that would outweigh any detriment.

This recommendation is reflected in the model legislative provisions in Appendix A.

The Panel also considers that the ACCC should be empowered to grant a block exemption in respect of specified conduct in particular market conditions. This would enable the ACCC to create 'safe harbours' for businesses where they engage in conduct that is unlikely to substantially lessen competition, and avoid the time and resources required to seek an authorisation or notification.

## Recommendation 39 — Block exemption power

A block exemption power, exercisable by the ACCC, should be introduced and operate alongside the authorisation and notification frameworks in Part VII of the CCA.

This power would enable the ACCC to create safe harbours, where conduct or categories of conduct are unlikely to raise competition concerns, on the same basis as the test proposed by the Panel for authorisations and notifications (see Recommendation 38).

The ACCC should also maintain a public register of all block exemptions, including those no longer in force. The decision to issue a block exemption would be reviewable by the Australian Competition Tribunal.

The Panel's recommended form of block exemption power is reflected in the model legislative provisions in Appendix A.

For further detail on authorisation, notification and block exemption, see Chapter 22.

## 3.15 ENFORCEMENT AND REMEDIES

The Panel supports the enforcement regime under the CCA, which confers both public and private enforcement rights in respect of the competition law.

In relation to public enforcement by the ACCC, there appears to be general approval of the severity of the sanctions for contravention of the competition law. However, the Panel agrees with the view of the ACCC that the current sanction for a corporation failing to comply with section 155 of the CCA is inadequate.

Compulsory evidence-gathering powers under section 155 of the CCA bolster the ACCC's ability to enforce the CCA. The Panel recommends that the fine a court may award for non-compliance with section 155 be increased to the same level as the fine for non-compliance with notice-based evidence-gathering powers in the *Australian Securities and Investments Commission Act 2001*. The ACCC should also be able to use section 155 to investigate possible contraventions of court-enforceable undertakings accepted by the ACCC under section 87B of the CCA.

Compulsory evidence-gathering powers can also impose a regulatory burden on recipients of compulsory notices. The Panel acknowledges concerns raised in submissions about the costs of compliance with section 155 notices issued by the ACCC. This is in part due to the increased use of technology leading to more electronic material being retained by businesses that may need to be searched in order to comply with a notice.

Means are available to reduce the regulatory burden associated with section 155 notices. First, the ACCC should accept a responsibility to frame section 155 notices in the narrowest form possible, consistent with the scope of the matter being investigated. Secondly, in complying with a section 155 notice, the recipient should be required to undertake a reasonable search, taking into account factors such as the number of documents involved and the ease and cost of retrieving the documents. That requirement could most effectively be introduced into the CCA by a statutory defence based on the criteria of a reasonable search.

#### Recommendation 40 — Section 155 notices

The section 155 power should be extended to cover the investigation of alleged contraventions of court-enforceable undertakings.

The ACCC should review its guidelines on section 155 notices having regard to the increasing burden imposed by notices in the digital age. Section 155 should be amended so that it is a defence to a 'refusal or failure to comply with a notice' under paragraph 155(5)(a) of the CCA that a recipient of a notice under paragraph 155(1)(b) can demonstrate that a reasonable search was undertaken in order to comply with the notice.

The fine for non-compliance with section 155 of the CCA should be increased in line with similar notice-based evidence-gathering powers in the *Australian Securities and Investments Commission Act 2001*.

Private enforcement of the competition law is an important right. However, there are many regulatory and practical impediments to the exercise of such a right. It is important to find ways to reduce those impediments.

Section 83 of the CCA is intended to facilitate private actions by enabling findings of fact made against a corporation in one proceeding (typically a proceeding brought by the ACCC) to be prima facie evidence against the corporation in another proceeding (typically a proceeding brought

by a private litigant). Many ACCC proceedings are resolved by a corporation making admissions of fact that establish the contravention, but it is uncertain whether section 83 applies to admissions as well as findings of fact.

The effectiveness of section 83 as a means of reducing the cost of private 'follow-on' proceedings would be enhanced if the section were amended to apply to admissions of fact made by a corporation in another proceeding, as well as findings of fact.

Concerns are expressed in submissions about the impact that extending section 83 to admissions of fact could have on the willingness of respondents to co-operate in cartel matters or settle matters with the ACCC, compromising the effectiveness of public enforcement of the CCA. The Panel doubts that this change to section 83 would materially alter the assessment by a respondent whether or not to settle an ACCC proceeding. Amongst other considerations, section 83 merely makes the admitted fact prima facie evidence of that fact in the follow-on proceeding. The respondent company remains free, should it so choose, to adduce evidence in the follow-on proceeding contrary to the admitted fact.

The proposed amendment to section 83 would remove doubt about its operation in the context of factual admissions and reduce the costs and risks of proceedings brought by persons who may have suffered loss and damage by reason of admitted contravening conduct.

#### Recommendation 41 — Private actions

Section 83 of the CCA should be amended so that it extends to admissions of fact made by the person against whom the proceedings are brought in addition to findings of fact made by the court.

This recommendation is reflected in the model legislative provisions in Appendix A.

In respect of contravening conduct that occurs overseas, a foreign corporation should be subject to Australian competition law regardless of whether it carries on business in Australia. Given that competition laws and policies are now commonplace around the world, there is no reason why private parties should have to seek ministerial consent before launching a proceeding that involves overseas conduct. This is addressed in Recommendation 26.

The Panel considers that small business needs greater assurance that competition complaints can be dealt with. Recommendation 53 deals with small business access to remedies.

For further detail on enforcement and remedies, see Chapter 23.

## 3.16 National Access Regime

The National Access Regime (contained in Part IIIA of the CCA) was originally established to enable third-party access to identified bottleneck infrastructure where it was apparent that economic efficiency would be enhanced by promoting competition in markets that were dependent upon access to that infrastructure.

The bottleneck infrastructure identified by the Hilmer Review included electricity wires, gas pipelines, telecommunication lines, freight rail networks, airports and ports. Distinct access regimes have emerged for these different types of infrastructure, reflecting their distinct physical, technical and economic characteristics. Those regimes appear to be achieving the original policy goals

identified by the Hilmer Review such that, today, Part IIIA plays only a limited role in regulating that bottleneck infrastructure.

However, the Panel acknowledges that Part IIIA continues to provide a legislative framework upon which industry-specific access regimes are based. Part IIIA is both a model and a 'backstop'. Accordingly, Part IIIA has an indirect role in supporting many industry-specific access regimes, even though its direct role is only limited.

The Panel has been told of the potential need for future access regulation of airport and port infrastructure. However, imposing an access regime upon privately developed single-user infrastructure is more likely to be inefficient than efficient, and impede the competitiveness of Australian industry.

The Panel agrees with the conclusion of the recent Productivity Commission (PC) inquiry that the National Access Regime is likely to generate net benefits to the community, but that its scope should be confined to ensure its use is limited to the exceptional cases, where the benefits arising from increased competition in dependent markets are likely to outweigh the costs of regulated third-party access.

In its report, the PC recommended the following changes to the declaration criteria in Part IIIA:

- that criterion (a) will be satisfied if access to an infrastructure service on reasonable terms and conditions through declaration (rather than access per se) would promote a material increase in competition in a dependent market;
- that criterion (b) will be satisfied where total foreseeable market demand for the infrastructure service over the declaration period could be met at least cost by the facility;
- as an alternative recommendation, that criterion (b) will be satisfied where it would be uneconomical for anyone (other than the service provider) to develop another facility to provide the service; and
- that criterion (f) will be satisfied if access on reasonable terms and conditions through declaration would promote the public interest.

The Panel agrees with the PC's proposed change to criterion (a), but considers that criterion (a) sets too low a threshold for declaration. The burdens of access regulation should not be imposed on the operations of a facility unless access is expected to produce efficiency gains from competition that are significant. This requires that competition be increased in a market that is significant and that the increase in competition be substantial.

The Panel supports the PC's alternative recommendation in respect of criterion (b). The alternative recommendation maintains the current language for criterion (b), while clarifying that duplication of the facility by the owner of the existing facility is not a relevant consideration.

As recently interpreted by the High Court in the Pilbara rail access case, criterion (b) asks a practical question whether it would be economically feasible, in other words profitable, for another facility to be developed — if it would, the facility is not a bottleneck. The Panel considers that maintaining the 'economically feasible' test for criterion (b) will best promote the competition policy objectives underpinning Part IIIA. Under that test, access regulation will only be considered where there is a bottleneck problem that needs to be addressed. Absent a bottleneck problem, competition and economic efficiency will be advanced if market participants are free to negotiate private arrangements concerning access.

The Panel considers that re-framing criterion (b) such that it requires an evaluation of whether a facility is a natural monopoly suffers from a number of shortcomings. These include that it can be trivially satisfied in the case of facilities that have been built with spare capacity and that it requires the decision-maker to evaluate least cost solutions in complex industries, burdened by information asymmetries where the risk of error is high.

The Panel supports the PC's recommendations in relation to criterion (f).

Decisions to declare a service under Part IIIA, or determine terms and conditions of access, are very significant economic decisions where the costs of making a wrong decision are likely to be high. The Panel favours empowering the Australian Competition Tribunal to undertake a merits review of access decisions, including hearing directly from employees of the business concerned and relevant experts where that would assist, while maintaining suitable statutory time limits for the review process.

## **Recommendation 42 — National Access Regime**

The declaration criteria in Part IIIA of the CCA should be targeted to ensure that third-party access only be mandated where it is in the public interest. To that end:

- Criterion (a) should require that access on reasonable terms and conditions through declaration promote a substantial increase in competition in a dependent market that is nationally significant.
- Criterion (b) should require that it be uneconomical for anyone (other than the service provider) to develop another facility to provide the service.
- Criterion (f) should require that access on reasonable terms and conditions through declaration promote the public interest.

The Competition Principles Agreement should be updated to reflect the revised declaration criteria.

The Australian Competition Tribunal should be empowered to undertake a merits review of access decisions, while maintaining suitable statutory time limits for the review process.

For further detail on the National Access Regime, see Chapter 24.

# 4 Institutions and Governance

## 4.1 A NATIONAL COMPETITION BODY

Several lessons may be drawn from Australia's experience of implementing the National Competition Policy (NCP):

- All jurisdictions need to commit to the policy and its implementation.
- Oversight of progress should be independent and transparent to 'hold governments to account'.
- The benefits of reform need to be argued and, where possible, measured.

Governance arrangements to implement reforms must be established in the context of Australia's federal structure. Many of the competition policy reforms outlined in this Report are overseen by state and territory governments. Although the Reform of the Federation White Paper may recommend changes to the way responsibilities are allocated across the Federation, it is reasonable to presume that all levels of government will continue to have a role in implementing competition policy reforms.

All Australian governments must have confidence in the governance arrangements for a reinvigorated round of competition policy reform to succeed.

The Panel believes that reinvigorating competition policy requires leadership from an institution specifically constituted for the purpose. Leadership encompasses advocacy for competition policy, driving implementation of the decisions made and conducting independent, transparent reviews of progress.

The National Competition Council (NCC), which oversaw the NCP, now has a considerably diminished role. It has been put to the Panel that the NCC no longer has the capacity to provide leadership in this domain. Recommendation 50 proposes that the remaining functions of the NCC, associated with the National Access Regime, be transferred to a new national access and pricing regulator. The NCC could then be dissolved.

The Productivity Commission (PC) is the only existing body with the necessary credibility and expertise to undertake this function, given its role as an independent research and advisory body on a range of economic, social and environmental issues affecting the welfare of Australians. But the PC's work is driven by the Australian Government and, if it were to have the competition policy function as well, its legislation and governance would need significant change.

The Australian Energy Market Commission (AEMC) is an example of an independent, national organisation, operating in an area of state government responsibility that has a governance structure supported by the Australian Government and the States and Territories. This is achieved through the AEMC's establishment under state legislation, which is then applied in other States and Territories and at the Commonwealth level. The national character of the organisation is further strengthened through the composition of the Commission itself, with state and territory Commissioners as well as a Commonwealth Commissioner.

The Panel considers that a new national competition body — the Australian Council for Competition Policy (ACCP) — should be established with a mandate to provide leadership and drive implementation of the evolving competition policy agenda.

The ACCP cannot be accountable to just one jurisdiction but must be accountable to them all. Similarly to the AEMC, it should be created by state and territory legislation applied by all participating jurisdictions. The ACCP should have a five-member board, consisting of two state and territory-nominated members and two members selected by the Australian Government, plus a Chair. Nomination of the Chair should rotate between the Australian Government and the States and Territories. The Chair should be appointed on a full-time basis and other members on a part-time basis.

Although members would be nominated and appointed by governments, their role should be to view competition policy from a national perspective and not to represent jurisdictional interests.

## Recommendation 43 — Australian Council for Competition Policy — Establishment

The National Competition Council should be dissolved and the Australian Council for Competition Policy (ACCP) established. Its mandate should be to provide leadership and drive implementation of the evolving competition policy agenda.

The ACCP should be established under legislation by one State and then by application in all other States and Territories and at the Commonwealth level. It should be funded jointly by the Australian Government and the States and Territories.

The ACCP should have a five-member board, consisting of two members nominated by state and territory Treasurers and two members selected by the Australian Government Treasurer, plus a Chair. Nomination of the Chair should rotate between the Australian Government and the States and Territories combined. The Chair should be appointed on a full-time basis and other members on a part-time basis.

Funding should be shared by all jurisdictions, with half of the funding provided by the Australian Government and half by the States and Territories in proportion to their population size.

## 4.2 FUNCTIONS OF THE NATIONAL BODY

The ACCP should have a broad role. In particular, the ACCP should advise governments on how to adapt competition policy to changing circumstances facing consumers and business. The ACCP should therefore develop an understanding of the state of competition across the Australian economy and report on it regularly.

The Panel sees advocacy for competition as a central function of the ACCP. Too often this has fallen by default to the Australian Competition and Consumer Commission (ACCC), which can be an uneasy role for a regulator to fulfil.

The ACCP should also act as an independent assessor of progress on reform, holding governments at all levels to account. Priority areas for reform identified in this Report could form an initial program of work for the ACCP.

#### Recommendation 44 — Australian Council for Competition Policy — Role

The Australian Council for Competition Policy should have a broad role encompassing:

- advocacy, education and promotion of collaboration in competition policy;
- independently monitoring progress in implementing agreed reforms and publicly reporting on progress annually;
- identifying potential areas of competition reform across all levels of government;
- making recommendations to governments on specific market design issues, regulatory reforms, procurement policies and proposed privatisations;
- undertaking research into competition policy developments in Australia and overseas; and
- ex-post evaluation of some merger decisions.

The effectiveness of the ACCP could be enhanced by assigning it a market studies function, which would create a consistent, effective and independent way for governments to seek advice and recommendations on recurrent and emerging competition policy issues.

Given the potential for conflicts between the ACCC's investigation and enforcement responsibilities and the scope of a market studies function, the Panel believes it is appropriate to vest such a power with the ACCP rather than the ACCC.

The market studies function would have a competition policy focus and complement, but not duplicate, the work of other bodies, such as the PC. For example, States and Territories could request the ACCP to undertake market studies of the provision of human services in their jurisdiction, as part of implementing the principles of choice and diversity of providers set out in Recommendation 2.

The use of mandatory information-gathering powers can help to ensure that a market study builds an accurate picture of the market but, on the other hand, may create an adversarial environment where participants show reluctance to co-operate and share information with the market studies body. The approach adopted by the PC — inviting interested parties to comment on issues and undertaking independent research, with mandatory legal powers as a backstop — appears to achieve desired outcomes.

For further detail on the establishment and functions of the ACCP, see Chapter 25.

## Recommendation 45 — Market studies power

The Australian Council for Competition Policy (ACCP) should have the power to undertake competition studies of markets in Australia and make recommendations to relevant governments on changes to regulation, or to the ACCC for investigation of potential breaches of the CCA.

The ACCP should have mandatory information-gathering powers to assist in its market studies function; however, these powers should be used sparingly.

The NCP recognised that different circumstances across the jurisdictions could lead to different approaches to either the scope or timing of reform. In agreeing with this approach, the Panel considers that the ACCP should be able to receive referrals from jurisdictions collectively as well as individually.

This would ensure that each jurisdiction has the freedom to identify its own concerns, while allowing the ACCP the flexibility to consider whether those concerns have broader or cross-jurisdictional impacts.

In addition, the Panel considers that all market participants, including small business and regulators, should have the opportunity to raise issues they would like to see become the subject of market studies. Funding could be set aside in the ACCP budget to undertake studies in addition to those referred by governments. The decision would rest with the ACCP as to which of these outside requests it might take up, and it would not be obliged to agree to all requests.

To give the ACCP the capacity to focus on the priorities of governments and market participants, the Ministerial Council on Federal Financial Relations would need to oversee priorities and resourcing.

#### Recommendation 46 — Market studies requests

All governments, jointly or individually, should have the capacity to issue a reference to the Australian Council for Competition Policy (ACCP) to undertake a competition study of a particular market or competition issue.

All market participants, including small business and regulators (such as the ACCC), should have the capacity to request market studies be undertaken by the ACCP.

The work program of the ACCP should be overseen by the Ministerial Council on Federal Financial Relations to ensure that resourcing addresses priority issues.

For further detail on market studies, see Section 25.6.

The competition policy environment is not static. New technologies can raise new issues and resolve older ones. The Panel considers that governments would benefit from an annual analysis of developments in the competition policy environment.

This would provide more detail on the specific priority issues or markets that should receive greater attention. It could also include recommending review mechanisms, particularly for more heavily regulated markets, to ensure more burdensome or intrusive regulatory frameworks remain fit for purpose.

Commenting on best practice and international developments would provide opportunities for governments to consider whether the outcomes of different approaches to reform in other jurisdictions apply within their own.

#### Recommendation 47 — Annual competition analysis

The Australian Council for Competition Policy should be required to undertake an annual analysis of developments in the competition policy environment, both in Australia and internationally, and identify specific issues or markets that should receive greater attention.

For further detail on competition analysis, see Section 25.9.

## 4.3 COMPETITION PAYMENTS

The Panel heard widespread support for the competition payments made by the Australian Government to state and territory governments under the NCP to recognise that the Australian Government received a disproportionate share of the increased revenue flowing from NCP reforms.

Although the quantum of the payments was not large compared to total state and territory revenues, the Panel consistently heard that their existence provided an additional argument that could be used to support reform. However, the Panel was also told that their effectiveness was limited by not being applied to the Australian Government nor consistently to local government.

On the other hand, as noted by the PC, a focus on payments and penalties 'has from time to time almost certainly misled the community as to the main rationale for reform ...'<sup>26</sup> This appears to underlie the observation, made by many stakeholders, that progress with competition policy reform waned once competition payments ceased.

That said, there is a case to be made that the benefits of reform, including any fiscal dividend, should be commensurate with the reform effort made. The differing revenue bases of the Commonwealth and the States and Territories mean that revenue may not flow in proportion to reform effort.

The PC should be tasked to undertake a study of reforms agreed to by the Australian Government and state and territory governments to estimate their effect on revenue in each jurisdiction. The ACCP could then assess whether reforms had been undertaken to a sufficient standard to warrant compensation payments. That assessment would be based on actual implementation of reforms, not on the basis of undertaking reviews or other processes.

#### Recommendation 48 — Competition payments

The Productivity Commission should be tasked to undertake a study of reforms agreed to by the Australian Government and state and territory governments to estimate their effect on revenue in each jurisdiction.

If disproportionate effects across jurisdictions are estimated, competition policy payments should ensure that revenue gains flowing from reform accrue to the jurisdictions undertaking the reform.

Reform effort should be assessed by the Australian Council for Competition Policy based on actual implementation of reform measures, not on undertaking reviews.

For further detail on competition payments, see Section 25.5.

## 4.4 COMPETITION AND CONSUMER REGULATOR

The Panel believes that enforcement of competition policy and enforcement of consumer protection matters are complementary and recommends both continue to be administered by one body.

Having a single body:

fosters a pro-market culture;

<sup>26</sup> Productivity Commission 2005, Review of National Competition Policy Reforms, Canberra, page 152.

- facilitates co-ordination and depth across the functions;
- provides a source of consistent information to business and consumers about their rights; and
- provides administrative savings and skills enhancement through pooling information, skills and expertise.

A single body also ensures that the issues of small business are not overlooked, as could be the case if the competition and consumer functions were separated into different bodies.

However, the Panel notes that tensions can also arise between the two functions, so it is important that the ACCC continues to maintain an appropriate balance between its competition-related regulatory tasks and its role in protecting consumers.

#### Recommendation 49 — ACCC functions

Competition and consumer functions should be retained within the single agency of the ACCC.

For further detail on ACCC functions, see Section 26.1

## 4.5 Access and Pricing Regulator

The Panel accepts that the functions of competition, consumer protection and economic regulation have synergies that can assist the ACCC to perform its functions and allow it to develop both wide and deep skills in understanding the operation of markets.

However, the culture and analytical approach required to regulate an industry differ from those typically characteristic of a competition law enforcement agency. There is also a risk that an industry regulator's views about the structure of a particular market could influence a merger decision.

The Panel therefore sees benefit in focusing the ACCC on its competition and consumer functions and separating out its current access and pricing functions into a separate, dedicated regulator. Amalgamating all Australian Government price regulatory functions into a single body will sharpen focus and strengthen analytical capacity in this important area of regulation.

The new body would subsume the access and pricing functions of the ACCC including: declaration and access arbitration functions under the telecommunications access regime in Part XIC of the *Competition and Consumer Act 2010* (CCA); price monitoring functions under the *Water Act 2007*; and access arbitration functions under the National Access Regime.

It would also include the functions of the Australian Energy Regulator (AER). The Panel notes strong support, especially in consultation with state governments, for energy regulation to be separated out from the ACCC. Including these functions in a new Access and Pricing Regulator would avoid the possibility of an industry-specific regulator being susceptible to 'capture' by the regulated industry. Therefore, the new body should not have responsibility for only one industry.

The proposed body would also take on the NCC's functions under the National Access Regime and under the National Gas Law, which would allow the NCC to be dissolved. This would result in the Access and Pricing Regulator undertaking both the declaration function under the National Access Regime and the current ACCC role in arbitrating the terms and conditions, where a facility is declared but terms and conditions are not able to be commercially negotiated.

The Panel does not foresee any conflict in a single regulator performing both functions and anticipates that there may be benefits. The Panel notes that, under the current telecommunications access regime (in Part XIC of the CCA), the ACCC performs both the declaration and arbitration functions.

The Access and Pricing Regulator could, over time, assume responsibility for other functions, if and when they were elevated into a national framework. For example, submissions propose the rail and water sectors as potential candidates for transfer, should States and Territories choose to do so.

#### **Recommendation 50 — Access and Pricing Regulator**

The following regulatory functions should be transferred from the ACCC and the NCC and be undertaken within a single national Access and Pricing Regulator:

- the telecommunications access and pricing functions of the ACCC;
- price regulation and related advisory roles of the ACCC under the Water Act 2007 (Cth);
- the powers given to the ACCC under the National Access Regime;
- the functions undertaken by the Australian Energy Regulator under the National Electricity Law, the National Gas Law and the National Energy Retail Law;
- the powers given to the NCC under the National Access Regime; and
- the powers given to the NCC under the National Gas Law.

Other consumer protection and competition functions should remain with the ACCC. Price monitoring and surveillance functions should also be retained by the ACCC.

The Access and Pricing Regulator should be constituted as a five-member board. The board should comprise two Australian Government-appointed members, two state and territory-nominated members and an Australian Government-appointed Chair. Two members (one Australian Government appointee and one state and territory appointee) should be appointed on a part-time basis.

Decisions of the Access and Pricing Regulator should be subject to review by the Australian Competition Tribunal.

The Access and Pricing Regulator should be established with a view to it gaining further functions if other sectors are transferred to national regimes.

For further detail on functions of the proposed Access and Pricing Regulator, see Chapter 27.

## 4.6 ACCC GOVERNANCE

The ACCC is established under the CCA as a statutory authority. It is governed by a Chairperson and other persons appointed as members of the Commission (usually called Commissioners). Decisions are made by the Chairperson and Commissioners meeting together (or as a division of the Commission), save where a power has been delegated to a member of the Commission. The Commission is assisted by its staff. The Chairperson and Commissioners are appointed on a full-time basis, resulting in their performing executive roles — although this isn't conferred by legislation.

The Panel considers that the ACCC is a well-regarded and effective body, but its performance would be strengthened by including a more diverse range of views and experience at the Commission level. This can be achieved by introducing part-time Commissioners whose commitments beyond the ACCC

— including, potentially, in business, consumer advocacy and academic roles — would broaden the Commission's perspective. The part-time Commissioners would, of necessity, be non-executive members of the Commission, standing apart from the agency's day-to-day operations.

The Panel recommends that half of the ACCC Commissioners be appointed on a part-time basis, that Deputy Chair positions be abolished and that the Chairperson be appointed on either a full-time or a part-time basis.

The Panel sees no need to continue sectoral Commissioner positions within the ACCC, noting that all Commissioners are required to exercise decision-making functions across the range of the ACCC's operations. Furthermore, under section 7 of the CCA, the Minister is already required to consider whether nominees have knowledge of, or experience in, consumer protection and small business matters for all potential appointments to the Commission. The Panel feels this is sufficient to ensure appropriate consideration of sectoral interests in appointments.

The ACCC should report regularly to a broad-based committee of the Parliament, such as the House of Representatives Standing Committee on Economics, to build profile and credibility for the agency as well as subjecting it to direct accountability to the Parliament.

#### **Recommendation 51 — ACCC governance**

Half of the ACCC Commissioners should be appointed on a part-time basis. This could occur as the terms of the current Commissioners expire, with every second vacancy filled with a part-time appointee. The Chair could be appointed on either a full-time or a part-time basis, and the positions of Deputy Chair should be abolished.

The Panel believes that current requirements in the CCA (paragraphs 7(3)(a) and 7(3)(b)) for experience and knowledge of small business and consumer protection, among other matters, to be considered by the Minister in making appointments to the Commission are sufficient to represent sectoral interests in ACCC decision-making.

Therefore, the Panel recommends that the further requirements in the CCA that the Minister, in making all appointments, be satisfied that the Commission has one Commissioner with knowledge or experience of small business matters (subsection 10(1B)) and one Commissioner with knowledge or experience of consumer protection matters (subsection 7(4)) be abolished.

The ACCC should report regularly to a broad-based committee of the Parliament, such as the House of Representatives Standing Committee on Economics.

For further detail on ACCC governance, see Section 26.2.

Some submissions criticise the ACCC's use of the media as undermining the perceived impartiality of the agency in undertaking enforcement action. Advocating for competition policy would become the responsibility of the new ACCP, if established, but the ACCC would continue to communicate with the public through the media, including explaining enforcement priorities, educating business about compliance, and publishing enforcement outcomes.

The Panel believes the ACCC should establish, publish and report against a Media Code of Conduct in line with the principles laid out in the Dawson Review. This should counter the perception of partiality on the part of the ACCC, especially in enforcement actions.

# Recommendation 52 — Media Code of Conduct

The ACCC should establish, publish and report against a Code of Conduct for its dealings with the media with the aim of strengthening the perception of its impartiality in enforcing the law. The Code of Conduct should be developed with reference to the principles outlined in the 2003 *Review of the Competition Provisions of the Trade Practices Act*.

For further detail on ACCC and the media, see Section 26.3.

# 5 SMALL BUSINESS

Small business makes a vital contribution to Australia's economy. The Panel has been especially mindful of the concerns and interests of small business in the context of the Review.

During the course of consultations, the Panel met in forums with more than 150 small businesses. These meetings supplemented written submissions made to the Review.

The issues raised in forums and submissions were broad-ranging, including: unequal bargaining power in dealing with larger businesses (including concerns about collective bargaining); the compliance burden of regulation; and difficulties in competing with (local) government-run enterprises, particularly where government is also the rule-maker.

This Report contains a number of recommendations that address these and other concerns of small business.

Specifically, the Panel proposes changes to strengthen the 'misuse of market power' provisions of the *Competition and Consumer Act 2010* (CCA) at Recommendation 30, and sets out its views on the unconscionable conduct provisions in Section 19.3. We also consider other issues affecting small business, such as standards (see Recommendation 11), licensing, planning and zoning (see Recommendation 9) and competitive neutrality (see Recommendations 15 - 17) elsewhere in this Report.

In this chapter, we consider access to remedies, collective bargaining and industry codes.

# 5.1 ACCESS TO REMEDIES

Submissions express concern that, for various reasons including resource priorities, the Australian Competition and Consumer Commission (ACCC) is unable to pursue all small business complaints. They further submit that small businesses either lack the time and financial resources to take action themselves or are concerned about the impact this might have on their ongoing business relationships.

The Panel notes the report of the Productivity Commission's (PC's) review of *Access to Justice Arrangements*, establishment of the Small Business and Family Enterprise Ombudsman, and the current proposal to extend unfair contract terms to small business contracts.

The Panel considers that small businesses need greater assurance that competition complaints can be dealt with. Understandably, the ACCC is not able to take proceedings in respect of all complaints brought to it. However, the ACCC should place some priority on its response to small business complaints concerning the competition law.

If the ACCC determines that it is unable to pursue a particular complaint on behalf of a small business, the ACCC must communicate clearly and promptly its reasons for not acting and direct the complainant to alternative dispute resolution schemes.

Where the ACCC considers a complaint has merit but is not a priority for public enforcement, it should take a more active role in connecting small business with dispute resolution schemes. The ACCC should also test the law on a regular basis to assure small business that the law is being enforced.

The Panel supports submissions' positive comments about the efficacy of the various state and Commonwealth small business commissioners, small business offices and ombudsmen services and does not consider that a separate tribunal is warranted to deal specifically with competition issues.

The Panel also endorses a number of recommendations contained in the PC's Access to Justice Arrangements report.

The Panel considers that, as implementation of a number of small business related recommendations do not require legislative change, consultation on these changes could commence following agreement by the Australian Government.

#### Recommendation 53 — Small business access to remedies

The ACCC should take a more active approach in connecting small business to alternative dispute resolution schemes where it considers complaints have merit but are not a priority for public enforcement.

Where the ACCC determines it is unable to pursue a particular complaint on behalf of a small business, the ACCC should communicate clearly and promptly its reasons for not acting and direct the business to alternative dispute resolution processes. Where the ACCC pursues a complaint raised by a small business, the ACCC should provide that business with regular updates on the progress of its investigation.

Resourcing of the ACCC should allow it to test the law on a regular basis to ensure that the law is acting as a deterrent to unlawful behaviour.

Small business commissioners, small business offices and ombudsmen should work with business stakeholder groups to raise awareness of their advice and dispute resolution services.

The Panel endorses the following recommendations from the Productivity Commission's *Access to Justice Arrangements* report:

- Recommendations 8.2 and 8.4 to ensure that small businesses in each Australian jurisdiction have access to effective and low cost small business advice and dispute resolution services;
- Recommendation 8.3 to ensure that small business commissioners, small business offices or ombudsmen provide a minimum set of services, which are delivered in an efficient and effective manner;
- Recommendation 9.3 to ensure that future reviews of industry codes consider whether dispute
  resolution services provided pursuant to an industry code, often by industry associations or
  third parties, are provided instead by the Australian Small Business Commissioner under the
  framework of that industry code;
- Recommendation 11.1 to broaden the use of the Federal Court's fast track model to facilitate lower cost and more timely access to justice; and
- Recommendation 13.3 to assist in managing the costs of litigation, including through the use of costs budgets for parties engaged in litigation.<sup>27</sup>

For further detail on small business access to remedies, see Chapter 23

<sup>27</sup> Productivity Commission 2014, Access to Justice Arrangements, Inquiry Report No. 72, Canberra.

## 5.2 COLLECTIVE BARGAINING

Submissions broadly support the exemption process for collective bargaining by small business, which is designed to recognise unequal bargaining power between parties to a business transaction. The process of exemption through notification should be capable of addressing a number of the issues raised by small businesses in their dealings with big businesses.

However, the provisions are not being used as frequently as they might. Various improvements could be made, including increasing the flexibility of collective bargaining and improving the framework as it relates to collective boycott activities. For example, one change is to enable the group of businesses covered by a notification to be altered without the need for a fresh notification to be filed.

Raising awareness of these provisions, including but not limited to raising awareness of co-operatives, will promote their use and potentially strengthen the bargaining position of small businesses dealing with large businesses.

## Recommendation 54 — Collective bargaining

The CCA should be reformed to introduce greater flexibility into the notification process for collective bargaining by small business.

Reform should include allowing:

- the nomination of members of the bargaining group, such that a notification could be lodged to cover future (unnamed) members;
- the nomination of the counterparties with whom the group seeks to negotiate, such that a notification could be lodged to cover multiple counterparties; and
- different timeframes for different collective bargaining notifications, based on the circumstances of each application.

Additionally, the ACCC should be empowered to impose conditions on notifications involving collective boycott activity, the timeframe for ACCC assessment of notifications for conduct that includes collective boycott activity should be extended from 14 to 60 days to provide more time for the ACCC to consult and assess the proposed conduct, and the ACCC should have a limited 'stop power' to require collective boycott conduct to cease, for use in exceptional circumstances where a collective boycott is causing imminent serious detriment to the public.

The current maximum value thresholds for a party to notify a collective bargaining arrangement should be reviewed in consultation with representatives of small business to ensure that they are high enough to include typical small business transactions.

The ACCC should take steps to enhance awareness of the exemption process for collective bargaining and how it might be used to improve the bargaining position of small businesses in dealings with large businesses. The ACCC should also amend its collective bargaining notification guidelines. This should include providing information about the range of factors considered relevant to determining whether a collective boycott may be necessary to achieve the benefits of collective bargaining.

For further detail on collective bargaining, see Section 22.2.

## 5.3 INDUSTRY CODES

Codes of conduct play an important role under the CCA by providing a flexible regulatory framework to set norms of behaviour. Codes of conduct complement the provisions of the CCA and generally apply to relationships between businesses within a particular industry. Codes also provide a mechanism to implement industry-specific dispute resolution frameworks.

The Panel notes that the CCA was recently amended to give the ACCC additional powers to issue infringement notices for alleged breaches of industry codes. The first code to incorporate the new civil penalties is the new Franchising Code of Conduct, which took effect from 1 January 2015. Experience with administering these new provisions is needed before determining whether they should be applied more broadly.

For further detail on industry codes, see Section 19.4.

## 5.4 COMPETITIVE NEUTRALITY

For many small businesses, competitive neutrality persists as an area of concern. Governments often have an undue advantage when they compete with small businesses, enabling them to penetrate markets more deeply and charge artificially lower prices than private sector competitors.

The Panel considers that transparency of current competitive neutrality arrangements should be improved and obligations on governments not to breach competitive neutrality principles should be strengthened. The Panel makes three recommendations in this regard (see Recommendations 15-17).

For further detail on competitive neutrality, see Chapter 13.

## 5.5 REGULATORY RESTRICTIONS

The ability of small businesses to compete will also be enhanced by a number of the Panel's recommendations to remove regulatory restrictions.

In particular, the Panel notes that recommendations concerning planning and zoning and a review of regulatory restrictions will assist small business (see Recommendations 8, 9 and 11).

For further detail on regulatory restrictions, see Chapter 10.

<sup>28</sup> Australian Competition and Consumer Commission 2015, New powers for ACCC will strengthen franchising industry, media release, 21 January, Sydney.

# 6 RETAIL MARKETS

Competition in retail markets has been an important focus for submissions and the Review. This includes issues relating to how competition is operating in grocery and fuel retailing, regulations on planning, zoning and trading hours, and specific regulations, such as those affecting pharmacy and liquor retailing.

Some of these issues are dealt with elsewhere in this Report, which includes specific recommendations on planning and zoning (see Recommendation 9) and pharmacy (see Recommendation 14). Retail liquor licensing should be reviewed as part of the general process of regulatory review (see Recommendation 8).

#### 6.1 SUPERMARKETS

A large number of submissions raise issues relating to supermarkets. However, on further investigation, most turn out to concern policy and legal issues that apply more broadly than just to supermarkets. Accordingly, many of the Panel's recommendations to deal with these broader issues also apply to supermarkets.

Some small supermarkets allege that the major supermarkets chains misuse their market power, including through 'predatory capacity' and targeting particular retailers. Suppliers also raise concerns about misuse of market power and unconscionable conduct by the major chains.

The Panel cannot adjudicate instances where breaches of the *Competition and Consumer Act 2010* (CCA) are alleged to have occurred but notes that the CCA generally prohibits conduct that harms the competitive process, not individual competitors.

The Panel recommends strengthening the misuse of market power provisions of the CCA at Recommendation 30. The current unconscionable conduct provisions appear to be working as intended to meet their policy goals, but active and ongoing review of these provisions should occur as matters progress before the courts. In this context, the Panel notes that in December 2014 the Federal Court, by consent, made declarations that Coles Supermarkets Australia Pty Ltd engaged in unconscionable conduct in 2011 in its dealings with certain suppliers in the supermarket sector. 29

Introducing a properly designed and effective industry code should also assist in ensuring that suppliers are able to contract fairly and efficiently. The Panel notes that the Australian Government has announced a Food and Grocery Code of Conduct, covering grocery suppliers and binding those retailers and wholesalers that agree to sign on to the Code.<sup>30</sup>

Removing barriers to entry and other regulatory barriers would strengthen competition in the supermarket sector. Planning and zoning restrictions are limiting the growth of new entrants such as ALDI and, as the ACCC has identified, more broadly affect the ability of independent supermarkets to

.

<sup>29</sup> Australian Competition and Consumer Commission v Coles Supermarkets Australia Pty Ltd [2014] FCA 1405 (22 December 2014)

<sup>30</sup> Billson, B (Minister for Small Business) 2015, Grocery Code to improve relationships between retailers, wholesalers and suppliers, media release 2 March, Canberra.

compete. The Panel recommends changes to address concerns about planning and zoning rules (see Recommendation 9).

Trading hours restrictions and restrictions preventing supermarkets from selling liquor impede competition. The Panel recommends that restrictions preventing supermarkets from selling liquor be reviewed as part of a new round of regulation reviews (see Recommendation 8) and that retail trading hours be deregulated (see Recommendation 12).

Supermarket operation has undergone a number of structural changes, including: greater vertical integration and use of 'home brands'; an increase in the range and categories of goods sold within supermarkets; and greater participation by supermarket operators in other sectors.

Like all structural changes, these can result in dislocation and other costs that affect the wellbeing of other parties. The move of larger supermarket chains into regional areas can also raise concerns about a loss of amenity and changes to the community.

While the Panel is sensitive to these concerns, they do not of themselves raise issues for competition policy or law.

For further detail on supermarkets, see Section 15.1.

#### 6.2 FUEL RETAILING

The Panel makes no specific recommendations in relation to fuel retailing, although a number of recommendations are relevant to submissions made in that context.

Petrol discount shopper dockets are a source of considerable concern, particularly for small competitors in the context of grocery and fuel markets. These discounts were up to 45 cents per litre<sup>31</sup> but are now limited to 4 cents per litre through undertakings to the ACCC.<sup>32</sup>

The Panel is not persuaded that consumers are made worse off by, rather than benefitting from, the availability of discounts at their current levels. The Panel notes the undertakings accepted by the ACCC. Further, the Panel recommends changes to the misuse of market power provisions of the CCA (see Recommendation 30), which should assist if future competition concerns emerge in this context.

Stakeholders express concerns that prices are higher in certain regional areas. On the information before it, the Panel does not consider that differences in pricing between regions are explained by any clear shortcoming in the competition law or policy. The Panel notes the 17 December 2014 Direction from the Minister for Small Business to the ACCC issued under the prices surveillance provisions of the CCA to monitor 'prices, costs and profits relating to the supply of unleaded petroleum products in the petroleum industry in Australia for three years'. <sup>33</sup> This will provide further information to assist in assessing any competition concerns in the sector, including in regional areas.

The Panel expresses no view as to the effect the Informed Sources pricing information sharing service has on competition. More generally, the Panel recommends that section 45 of the CCA be extended

<sup>31</sup> Sims, R 2013 *Thoughts on market concentration issues* speech to the Australian Food and Grocery Council Industry Leaders Forum, Canberra, 30 October.

<sup>32</sup> Australian Competition and Consumer Commission 2013, *Coles and Woolworths undertake to cease supermarket subsidised fuel discounts*, media release 6 December, Canberra.

<sup>33</sup> Explanatory Statement, Competition and Consumer Act 2010 — Monitoring of the Prices, Costs and Profits Relating to the Supply of Unleaded Petroleum Products in the Petroleum Industry in Australia (09/12/2014).

to cover concerted practices which have the purpose, effect or likely effect of substantially lessening competition.

Submissions raise concerns about the New South Wales Government mandate requiring that a certain proportion of petrol sold in that State contain ethanol. The Panel considers that this mandate should be reviewed as part of the proposed new round of regulation review (see Recommendation 8), and repealed unless it can be demonstrated that the benefits of the restriction to the community as a whole outweigh the costs, and the objectives of the policy can only be achieved by restricting competition.

In relation to the regulation of petrol price display boards, the Panel considers that the case for wider regulation to require the undiscounted price (only) to be displayed has not been made. The Panel notes that differences in regulations across jurisdictions create a 'natural experiment', which will provide evidence to assist Ministers in determining whether these regulations have any effect on competition and whether they are in the public interest.

In relation to proposals to introduce a national scheme based on Fuelwatch in Western Australia, the Panel considers that further evidence, both of a problem needing to be addressed and of the benefits and costs of addressing it in this way, would be necessary before making any decision to proceed.

For further detail on fuel retailing, see Section 15.2.

# 7 IMPLEMENTATION

Reforming Australia's competition policy, laws and institutions represents an ambitious agenda, which will require action by all levels of government. Although some recommendations can be implemented by jurisdictions acting independently, the Panel believes outcomes will be enhanced through co-operation between governments. Competition reform will have economy-wide impacts and therefore merits national action.

To commence consideration of a national competition reform agenda, this Report should be discussed with state and territory governments as soon as practicable. This will allow all governments to make considered responses, including identifying aspects of the agenda where they see value in collaboration.

# **Recommendation 55** — Implementation

The Australian Government should discuss this Report with the States and Territories as soon as practicable following its receipt.

Recommendation 48 is that the Productivity Commission (PC) be tasked with modelling the revenue effects in each jurisdiction of reforms agreed by governments in the wake of this Review. However, prior to that modelling exercise, the Panel believes that governments would benefit from modelling the economic effects of the recommendations in this Review. This modelling will assist governments in determining the gains from proposals and the prioritisation of reforms.

## Recommendation 56 — Economic modelling

The Productivity Commission should be tasked with modelling the recommendations of this Review as a package (in consultation with jurisdictions) to support discussions on policy proposals to pursue.

A 'road map' in Section 29.3 illustrates recommendations that can be implemented by different levels of government. For further detail on implementation see Part 6.